

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Andersen, Torben M.

Working Paper

Increasing longevity and social security reforms

CESifo Working Paper, No. 1789

Provided in Cooperation with:

Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Andersen, Torben M. (2006): Increasing longevity and social security reforms, CESifo Working Paper, No. 1789, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/25833

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



PANELS WITH NONSTATIONARY MULTIFACTOR ERROR STRUCTURES

GEORGE KAPETANIOS M. HASHEM PESARAN TAKASHI YAMAGATA

CESIFO WORKING PAPER NO. 1788
CATEGORY 10: EMPIRICAL AND THEORETICAL METHODS
AUGUST 2006

An electronic version of the paper may be downloaded
 from the SSRN website: www.SSRN.com
 from the RePEc website: www.RePEc.org
 from the CESifo website: www.CESifo-group.de

PANELS WITH NONSTATIONARY MULTIFACTOR ERROR STRUCTURES

Abstract

The presence of cross-sectionally correlated error terms invalidates much inferential theory of panel data models. Recently work by Pesaran (2006) has suggested a method which makes use of cross-sectional averages to provide valid inference for stationary panel regressions with multifactor error structure. This paper extends this work and examines the important case where the unobserved common factors follow unit root processes and could be cointegrated. It is found that the presence of unit roots does not affect most theoretical results which continue to hold irrespective of the integration and the cointegration properties of the unobserved factors. This finding is further supported for small samples via an extensive Monte Carlo study. In particular, the results of the Monte Carlo study suggest that the cross-sectional average based method is robust to a wide variety of data generation processes and has lower biases than all of the alternative estimation methods considered in the paper.

JEL Code: C12, C13, C33.

Keywords: cross section dependence, large panels, unit roots, principal components, common correlated effects.

George Kapetanios
Department of Economics
Queen Mary, University of London
Mile End Road
London E1 4NS
United Kingdom
g.kapetanios@gmul.ac.uk

M. Hashem Pesaran Faculty of Economics Cambridge University Cambridge, CB3 9DD United Kingdom mhp1@econ.cam.ac.uk

Takashi Yamagata
Faculty of Economics
Cambridge University
Cambridge, CB3 9DD
United Kingdom
takashi.yamagata@econ.cam.ac.uk

July 31, 2006

The authors thank Vanessa Smith and Elisa Tosetti for helpful comments on a preliminary version of this paper. Hashem Pesaran and Takashi Yamagata acknowledge financial support from the ESRC (Grant No. RES-000-23-0135).

1 Introduction

Panel data sets have been increasingly used in economics to analyze complex economic phenomena. One of their attractions is the ability to use an extended data set to obtain information about parameters of interest which are assumed to have common values across panel units. Most of the work carried out on panel data has usually assumed some form of cross sectional independence to derive the theoretical properties of various inferential procedures. However, such assumptions are often suspect and as a result recent advances in the literature have focused on estimation of panel data models subject to error cross sectional dependence.

A number of different approaches have been advanced for this purpose. In the case of spatial data sets where a natural immutable distance measure is available the dependence is often captured through "spatial lags" using techniques familiar from the time series literature. In economic applications, spatial techniques are often adapted using alternative measures of "economic distance". This approach is exemplified in work by Lee and Pesaran (1993), Conley and Dupor (2003), Conley and Topa (2002) and Pesaran, Schuermann, and Weiner (2004), as well as the literature on spatial econometrics recently surveyed by Anselin (2001). In the case of panel data models where the cross section dimension (N) is small (typically N < 10) and the time series dimension (T) is large the standard approach is to treat the equations from the different cross section units as a system of seemingly unrelated regression equations (SURE) and then estimate the system by the Generalized Least Squares (GLS) techniques.

In the case of panels with large cross section dimension, SURE approach is not practical and has led a number of investigators to consider unobserved factor models, where the cross section error correlations are defined in terms of the factor loadings. Use of factor models is not new in economics and dates back to the pioneering work of Stone (1947) who applied the principal components (PC) analysis of Hotelling to US macroeconomic time series over the period 1922-1938 and was able to demonstrate that three factors (namely total income, its rate of change and a time trend) explained over 97 per cent of the total variations of all the 17 macro variables that he had considered. Until recently, subsequent applications of the PC approach to economic times series has been primarily in finance. See, for example, Chamberlain and Rothschild (1983), Connor and Korajzcyk (1986) and Connor and Korajzcyk (1988). But more recently the unobserved factor models have gained popularity for forecasting with a large number of variables as advocated by Stock and Watson (2002).

The factor model is used very much in the spirit of the original work by Stone, in order to summarize the empirical content of a large number of macroeconomics variables by a small set of factors which, when estimated using principal components, is then used for further modelling and/or forecasting. A related literature on dynamic factor models has also been put forward by Forni and Reichlin (1998) and Forni, Hallin, Lippi, and Reichlin (2000).

Recent uses of factor models in forecasting focuses on consistent estimation of unobserved factors and their loadings. Related theoretical advances by Bai and Ng (2002) and Bai (2003) are also concerned with estimation and selection of unobserved factors and do not consider the estimation and inference problems in standard panel data models where the objects of interest are slope coefficients of the conditioning variables (regressors). In such panels the unobserved factors are viewed as nuisance variables, introduced primarily to model the cross section dependencies of the error terms in a parsimonious manner relative to the SURE formulation.

Despite these differences knowledge of factor models could still be useful for the analysis of panel data models if it is believed that the errors might be cross sectionally correlated. Disregarding the possible factor structure of the errors in panel data models can lead to inconsistent parameter estimates and incorrect inference. Coakley, Fuertes, and Smith (2002) suggest a possible solution to the problem using the method of Stock and Watson (2002). But, as Pesaran (2006) shows, the PC approach proposed by Coakley, Fuertes, and Smith (2002) can still yield inconsistent estimates. Pesaran (2006) suggests a new approach by noting that linear combinations of the unobserved factors can be well approximated by cross section averages of the dependent variable and the observed regressors. This leads to a new set of estimators, referred to as the Common Correlated Effects estimators, that can be computed by running standard panel regressions augmented with the cross section averages of the dependent variables. The CCE procedure is applicable to panels with a single or multiple unobserved factors so long as the number of unobserved factors is fixed.

In this paper we extend the analysis of Pesaran (2006) to the case where one or more of the unobserved common factors could be integrated of order 1, or I(1). This also allows for the possibility of cointegration amongst the I(1) factors and does not require an a priori knowledge of the number of unobserved factors, or their integration or cointegration properties. It is only required that the number of unobserved factors remain fixed as the sample size is increased. The theoretical findings of the paper are further supported for small samples via an extensive Monte Carlo study. In particular, the results of the Monte Carlo study

clearly show that the CCE estimator is robust to a wide variety of data generation processes and has lower biases than all of the alternative estimation methods considered in the paper.

The structure of the paper is as follows: Section 2 provides an overview of the method suggested by Pesaran (2006) in the case of stationary factor processes. Section 3 provides the theoretical framework of the analysis of nonstationarity. In this section the theoretical properties of the various estimators are presented. Section 4 presents an extensive Monte Carlo study, and Section 5 concludes.

2 Panel Data Models with Observed and Unobserved Common Effects

In this section we review the methodology introduced in Pesaran (2006). Let y_{it} be the observation on the i^{th} cross section unit at time t for i = 1, 2, ..., N; t = 1, 2, ..., T, and suppose that it is generated according to the following linear heterogeneous panel data model

$$y_{it} = \alpha_i' \mathbf{d}_t + \beta_i' \mathbf{x}_{it} + \gamma_i' \mathbf{f}_t + \varepsilon_{it}, \tag{1}$$

where \mathbf{d}_t is a $n \times 1$ vector of observed common effects, which is partitioned as $\mathbf{d}_t = (\mathbf{d}'_{1t}, \mathbf{d}'_{2t}, \mathbf{d}'_{3t})'$ where \mathbf{d}_{1t} is a $n_1 \times 1$ vector of deterministic components such as intercepts or seasonal dummies, \mathbf{d}_{2t} is a $n_2 \times 1$ vector of unit root stochastic observed common effects and \mathbf{d}_{3t} is a $n_3 \times 1$ vector of stationary stochastic observed common effects, with $n = n_1 + n_2 + n_3$, \mathbf{x}_{it} is a $k \times 1$ vector of observed individual-specific regressors on the i^{th} cross section unit at time t, \mathbf{f}_t is the $m \times 1$ vector of unobserved common effects, and ε_{it} are the individual-specific (idiosyncratic) errors assumed to be independently distributed of $(\mathbf{d}_t, \mathbf{x}_{it})$. The unobserved factors, \mathbf{f}_t , could be correlated with $(\mathbf{d}_t, \mathbf{x}_{it})$, and to allow for such a possibility the following specification for the individual specific regressors will be considered

$$\mathbf{x}_{it} = \mathbf{A}_i' \mathbf{d}_t + \mathbf{\Gamma}_i' \mathbf{f}_t + \mathbf{v}_{it}, \tag{2}$$

where \mathbf{A}_i and $\mathbf{\Gamma}_i$ are $n \times k$ and $m \times k$, factor loading matrices with fixed components, \mathbf{v}_{it} are the specific components of \mathbf{x}_{it} distributed independently of the common effects and across i, but assumed to follow general covariance stationary processes. In this paper we allow for one or more of the common factors, \mathbf{d}_t and \mathbf{f}_t , to be I(1).

Combining (1) and (2) we now have

$$\mathbf{z}_{it} = \begin{pmatrix} y_{it} \\ \mathbf{x}_{it} \end{pmatrix} = \mathbf{B}'_{i} \mathbf{d}_{t} + \mathbf{C}'_{i} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

$${}_{(k+1)\times n} \mathbf{n}^{\times 1} + \mathbf{u}_{it} \mathbf{f}_{t} + \mathbf{u}_{it},$$

where

$$\mathbf{u}_{it} = \begin{pmatrix} \varepsilon_{it} + \boldsymbol{\beta}_i' \mathbf{v}_{it} \\ \mathbf{v}_{it} \end{pmatrix}, \tag{4}$$

$$\mathbf{B}_{i} = \begin{pmatrix} \boldsymbol{\alpha}_{i} & \mathbf{A}_{i} \end{pmatrix} \begin{pmatrix} \mathbf{I}_{n} & \mathbf{0} \\ \boldsymbol{\beta}_{i} & \mathbf{I}_{k} \end{pmatrix}, \mathbf{C}_{i} = \begin{pmatrix} \boldsymbol{\gamma}_{i} & \boldsymbol{\Gamma}_{i} \end{pmatrix} \begin{pmatrix} \mathbf{I}_{m} & \mathbf{0} \\ \boldsymbol{\beta}_{i} & \mathbf{I}_{k} \end{pmatrix}, \tag{5}$$

 \mathbf{I}_k is an identity matrix of order k, and the rank of \mathbf{C}_i is determined by the rank of the $m \times (k+1)$ matrix of the unobserved factor loadings

$$\tilde{\Gamma}_i = (\gamma_i \quad \Gamma_i). \tag{6}$$

As discussed in Pesaran (2006), the above set up is sufficiently general and renders a variety of panel data models as special cases. In the panel literature with T small and N large, the primary parameters of interest are the means of the individual specific slope coefficients, β_i , i = 1, 2, ..., N. The common factor loadings, α_i and γ_i , are generally treated as nuisance parameters. In cases where both N and T are large, it is also possible to consider consistent estimation of the factor loadings. The presence of unobserved factors in (1) implies that estimation of β_i and its cross sectional mean cannot be undertaken using standard methods. Pesaran (2006) has suggested using cross section averages of y_{it} and \mathbf{x}_{it} as proxies for the unobserved factors in (1). To see why such an approach could work, consider simple cross section averages of the equations in (3)¹

$$\bar{\mathbf{z}}_t = \bar{\mathbf{B}}' \mathbf{d}_t + \bar{\mathbf{C}}' \mathbf{f}_t + \bar{\mathbf{u}}_t, \tag{7}$$

where

$$\bar{\mathbf{z}}_t = \frac{1}{N} \sum_{i=1}^{N} \mathbf{z}_{it}, \ \bar{\mathbf{u}}_t = \frac{1}{N} \sum_{i=1}^{N} \mathbf{u}_{it},$$

and

$$\bar{\mathbf{B}} = \frac{1}{N} \sum_{i=1}^{N} \mathbf{B}_i, \ \bar{\mathbf{C}} = \frac{1}{N} \sum_{i=1}^{N} \mathbf{C}_i.$$
 (8)

Suppose that

$$Rank(\mathbf{\bar{C}}) = m \le k+1, \text{ for all } N.$$
 (9)

Then, we have

$$\mathbf{f}_{t} = \left(\bar{\mathbf{C}}\bar{\mathbf{C}}'\right)^{-1}\bar{\mathbf{C}}\left(\bar{\mathbf{z}}_{t} - \bar{\mathbf{B}}'\mathbf{d}_{t} - \bar{\mathbf{u}}_{t}\right). \tag{10}$$

But since

$$\mathbf{\bar{u}}_t \stackrel{q.m.}{\to} \mathbf{0}$$
, as $N \to \infty$, for each t , (11)

and

$$\mathbf{\bar{C}} \stackrel{p}{\to} \mathbf{C} = \mathbf{\tilde{\Gamma}} \begin{pmatrix} \mathbf{I}_m & \mathbf{0} \\ \boldsymbol{\beta} & \mathbf{I}_k \end{pmatrix}, \text{ as } N \to \infty,$$
 (12)

¹Pesaran (2006) considers cross section weighted averages that are more general. But to simplify the exposition we confine our discussion to simple averages throughout.

where

$$\tilde{\Gamma} = (E(\gamma_i), E(\Gamma_i)) = (\gamma, \Gamma). \tag{13}$$

It follows, assuming that $Rank(\tilde{\Gamma}) = m$, that

$$\mathbf{f}_t - (\mathbf{C}\mathbf{C}')^{-1} \mathbf{C} \left(\bar{\mathbf{z}}_t - \bar{\mathbf{B}}' \mathbf{d}_t \right) \stackrel{p}{\to} \mathbf{0}, \text{ as } N \to \infty.$$

This suggests using $\bar{\mathbf{h}}_t = (\mathbf{d}_t', \bar{\mathbf{z}}_t')'$ as observable proxies for \mathbf{f}_t , and is the basic insight that lies behind the Common Correlated Effects estimators developed in Pesaran (2006). It is further shown that the CCE estimation procedure in fact holds even if $\tilde{\mathbf{\Gamma}}$ turns out to be rank deficient.

We now discuss the two estimators for the means of the individual specific slope coefficients proposed by Pesaran (2006). One is the Mean Group (MG) estimator proposed in Pesaran and Smith (1995) and the other is a generalization of the fixed effects estimator that allows for the possibility of cross section dependence. The former is referred to as the "Common Correlated Effects Mean Group" (CCEMG) estimator, and the latter as the "Common Correlated Effects Pooled" (CCEP) estimator.

The CCEMG estimator is a simple average of the individual CCE estimators, $\hat{\mathbf{b}}_i$ of $\boldsymbol{\beta}_i$,

$$\hat{\mathbf{b}}_{MG} = N^{-1} \sum_{i=1}^{N} \hat{\mathbf{b}}_{i}. \tag{14}$$

where

$$\hat{\mathbf{b}}_i = (\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i)^{-1} \mathbf{X}_i' \bar{\mathbf{M}} \mathbf{y}_i, \tag{15}$$

 $\mathbf{X}_i = (\mathbf{x}_{i1}, \mathbf{x}_{i2}, ..., \mathbf{x}_{iT})', \ \mathbf{y}_i = (y_{i1}, y_{i2}, ..., y_{iT})', \ \mathbf{\bar{M}}$ is defined by

$$\bar{\mathbf{M}} = \mathbf{I}_T - \bar{\mathbf{H}} \left(\bar{\mathbf{H}}' \bar{\mathbf{H}} \right)^{-1} \bar{\mathbf{H}}', \tag{16}$$

 $\bar{\mathbf{H}} = (\mathbf{D}, \bar{\mathbf{Z}}), \mathbf{D}$ and $\bar{\mathbf{Z}}$ being, respectively, the $T \times n$ and $T \times (k+1)$ matrices of observations on \mathbf{d}_t and $\bar{\mathbf{z}}_t$.

Efficiency gains from pooling of observations over the cross section units can be achieved when the individual slope coefficients, β_i , are the same. Such a pooled estimator of β , denoted by CCEP, is given by

$$\hat{\mathbf{b}}_{P} = \left(\sum_{i=1}^{N} \mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{X}_{i}\right)^{-1} \sum_{i=1}^{N} \mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{y}_{i}.$$
(17)

3 Theoretical Properties of CCE Estimators in Nonstationary Panel Data Models

The following assumptions will be used in the derivation of the asymptotic properties of the CCE estimators.

Assumption 1 (non-stationary common effects): The $(n_2 + m) \times 1$ vector of stochastic common effects, $\mathfrak{g}_t = (\mathbf{d}'_{2t}, \mathbf{f}'_t)'$, is a multivariate unit root process given by

$$\mathfrak{g}_t = \mathfrak{g}_{t-1} + \boldsymbol{\zeta}_{\mathfrak{g}t}$$

where $\zeta_{\mathfrak{g}t}$ is a $(n_2+m)\times 1$ vector of L_2 stationary near epoque dependent (NED) processes of size 1/2, distributed independently of the individual-specific errors, $\varepsilon_{it'}$ and $\mathbf{v}_{it'}$ for all i, t and t'.

Assumption 2 (stationary common effects): The $n_3 \times 1$ vector of common effects, \mathbf{d}_{3t} , is a covariance stationary process given by

$$\mathbf{d}_{3t} = \sum_{\ell=0}^{\infty} \mathbf{J}_{\ell} \boldsymbol{\psi}_{t-\ell},\tag{18}$$

where the matrices J_{ℓ} satisfy the condition

$$\sum_{\ell=0}^{\infty} \ell^s ||\mathbf{J}_{\ell}|| < \infty, \tag{19}$$

for some $s \ge 1/2$, and the ψ_t are distributed independently of the individual-specific errors, $\varepsilon_{it'}$ and $\mathbf{v}_{it'}$ for all i, t and t'.

Assumption 3 (individual-specific errors): The individual specific errors ε_{it} and \mathbf{v}_{jt} are distributed independently for all i, j and t. For each i, ε_{it} is serially uncorrelated with mean zero, a finite variance $\sigma_i^2 < K$, and a finite fourth-order cumulant. \mathbf{v}_{it} follows a linear stationary process with absolute summable autocovariances given by

$$\mathbf{v}_{it} = \sum_{\ell=0}^{\infty} \mathbf{S}_{i\ell} \boldsymbol{\nu}_{i,t-\ell}, \tag{20}$$

where ν_{it} are $k \times 1$ vectors of identically, independently distributed (IID) random variables with mean zero, the variance matrix, \mathbf{I}_k , and finite fourth-order cumulants. In particular, the $k \times k$ coefficient matrices $\mathbf{S}_{i\ell}$ satisfy the condition

$$\sum_{\ell=0}^{\infty} \ell^s ||\mathbf{S}_{i\ell}|| < \infty, \tag{21}$$

for all i and some $s \ge 1/2$.

Assumption 4 (factor loadings): The unobserved factor loadings, γ_i and Γ_i , are independently and identically distributed across i, and of the individual specific errors, ε_{jt} and \mathbf{v}_{jt} , the common factors, $\mathbf{g}_t = (\mathbf{d}'_{2t}, \mathbf{f}'_t)$, for all i, j and t with fixed means γ and Γ , respectively, and finite variances. In particular,

$$\gamma_i = \gamma + \eta_i, \ \eta_i \backsim IID \ (\mathbf{0}, \Omega_{\eta}), \ \text{for } i = 1, 2, ..., N,$$
 (22)

where Ω_{η} is a $m \times m$ symmetric non-negative definite matrix, and $\|\gamma\| < K$, $\|\Gamma\| < K$, and $\|\Omega_{\eta}\| < K$.

Assumption 5 (random slope coefficients): The slope coefficients, β_i , follow the random coefficient model

$$\boldsymbol{\beta}_i = \boldsymbol{\beta} + \boldsymbol{v}_i, \ \boldsymbol{v}_i \backsim IID \ (\mathbf{0}, \boldsymbol{\Omega}_{\boldsymbol{v}}), \text{ for } i = 1, 2, ..., N,$$
 (23)

where $\|\boldsymbol{\beta}\| < K$, $\|\boldsymbol{\Omega}_{\boldsymbol{v}}\| < K$, $\boldsymbol{\Omega}_{\boldsymbol{v}}$ is a $k \times k$ symmetric non-negative definite matrix, and the random deviations, \boldsymbol{v}_i , are distributed independently of $\boldsymbol{\gamma}_j$, $\boldsymbol{\Gamma}_j$, ε_{jt} , \boldsymbol{v}_{jt} , and \boldsymbol{g}_t for all i, j and t.

Assumption 6: (identification of β_i and β): Consider the cross section averages of the individual specific variables, \mathbf{z}_{it} , defined by $\bar{\mathbf{z}}_t = \frac{1}{N} \sum_{j=1}^{N} \mathbf{z}_{jt}$, and let

$$\bar{\mathbf{M}} = \mathbf{I}_T - \bar{\mathbf{H}} \left(\bar{\mathbf{H}}' \bar{\mathbf{H}} \right)^- \bar{\mathbf{H}}', \tag{24}$$

and

$$\mathbf{M}_g = \mathbf{I}_T - \mathbf{G} \left(\mathbf{G}' \mathbf{G} \right)^- \mathbf{G}', \tag{25}$$

where $\mathbf{G} = (\mathbf{D}, \mathbf{F})$, $\mathbf{D} = (\mathbf{d}_1, \mathbf{d}_2, ..., \mathbf{d}_T)'$, $\mathbf{F} = (\mathbf{f}_1, \mathbf{f}_2, ..., \mathbf{f}_T)'$ are $T \times n$ and $T \times m$ data matrices on observed and unobserved common factors, respectively, $\mathbf{\bar{Z}} = (\mathbf{\bar{z}}_1, \mathbf{\bar{z}}_2, ..., \mathbf{\bar{z}}_T)'$ is the $T \times (k+1)$ matrix of observations on the cross section averages, and $(\mathbf{\bar{H}'\bar{H}})^-$ and $(\mathbf{G'G})^-$ denote the generalized inverses of $\mathbf{\bar{H}'\bar{H}}$ and $\mathbf{G'G}$, respectively. Also denote the $T \times k$ observation matrix on individual specific regressors by $\mathbf{X}_i = (\mathbf{x}_{i1}, \mathbf{x}_{i2}, ..., \mathbf{x}_{iT})'$.

6a: (identification of $\boldsymbol{\beta}_i$): The $k \times k$ matrices $\hat{\boldsymbol{\Psi}}_{iT} = T^{-1} \left(\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i \right)$ and $\boldsymbol{\Psi}_{ig} = T^{-1} \left(\mathbf{X}_i' \mathbf{M}_g \mathbf{X}_i \right)$ are non-singular and $\hat{\boldsymbol{\Psi}}_{iT}^{-1}$ and $\boldsymbol{\Psi}_{ig}^{-1}$ have finite second order moments, for all i.

6b: (identification of β): The $k \times k$ pooled observation matrix $\hat{\Psi}_{NT}$ defined by

$$\hat{\mathbf{\Psi}}_{NT} = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{X}_{i}}{T} \right)$$
 (26)

is non-singular.

Remark 1 Note that Assumption 3 is slightly stronger than Assumption 2 of Pesaran (2006) which requires that

$$Var\left(\mathbf{v}_{it}\right) = \sum_{\ell=0}^{\infty} \mathbf{S}_{i\ell} \mathbf{S}'_{i\ell} = \mathbf{\Sigma}_{i} \le \mathbf{K} < \infty, \tag{27}$$

for all i and some constant matrix \mathbf{K} , where Σ_i is a positive definite matrix, rather than (21). Further note that (21) implies that \mathbf{v}_{it} are L_2 stationary near epoque dependent processes of size 1/2.

Remark 2 Assumption 1 implicitly allows for cointegration among the m unobserved factors. To see this it is useful to impose some further structure on $\zeta_{2\mathfrak{g}t}$ where $\zeta_{\mathfrak{g}t} = (\zeta'_{1\mathfrak{g}t}, \zeta'_{2\mathfrak{g}t})'$ is a partition of $\zeta_{\mathfrak{g}t}$ comformable to the partition of \mathfrak{g}_t in terms of d_{2t} and f_t . If we let $\zeta_{2\mathfrak{g}t} = \sum_{\ell=0}^{\infty} \mathbf{W}_{\ell} \varpi_{t-\ell}$ where $\varpi_{t-\ell}$ are i.i.d. with finite variance and $\sum_{\ell=0}^{\infty} \ell ||\mathbf{W}_{\ell}|| < \infty$ then if $\sum_{\ell=0}^{\infty} \mathbf{W}_{\ell}$ is of reduced rank equal to r, then $\zeta_{2\mathfrak{g}t}$ exhibits cointegration. Further f_t may be represented by a set of r stationary components and m-r random walk components. This representation will be of use in the proofs of our results. Note that the above MA representation of $\zeta_{2\mathfrak{g}t}$ implies that $\zeta_{2\mathfrak{g}t}$ is a vector of L_2 stationary near epoque dependent (NED) processes of size 1 and hence satisfy assumption 1^2 .

For each i and t = 1, 2, ..., T, writing the model in matrix notation we have

$$\mathbf{y}_i = \mathbf{D}\boldsymbol{\alpha}_i + \mathbf{X}_i\boldsymbol{\beta}_i + \mathbf{F}\boldsymbol{\gamma}_i + \boldsymbol{\varepsilon}_i, \tag{28}$$

where $\boldsymbol{\varepsilon}_i = (\varepsilon_{i1}, \varepsilon_{i2}, ..., \varepsilon_{iT})'$, and as set out in Assumption 5, $\mathbf{D} = (\mathbf{d}_1, \mathbf{d}_2, ..., \mathbf{d}_T)'$ and $\mathbf{F} = (\mathbf{f}_1, \mathbf{f}_2, ..., \mathbf{f}_T)'$. Using (28) in (15) we have

$$\hat{\mathbf{b}}_{i} - \boldsymbol{\beta}_{i} = \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T}\right)^{-1} \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{F}}{T}\right) \boldsymbol{\gamma}_{i} + \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T}\right)^{-1} \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\varepsilon_{i}}{T}\right), \tag{29}$$

which shows the direct dependence of $\hat{\mathbf{b}}_i$ on the unobserved factors through $T^{-1}\mathbf{X}_i'\mathbf{M}\mathbf{F}$. To examine the properties of this component, writing (2) and (7) in matrix notations, we first note that

$$\mathbf{X}_i = \mathbf{G}\mathbf{\Pi}_i + \mathbf{V}_i,\tag{30}$$

and

$$\bar{\mathbf{H}} = \mathbf{G}\bar{\mathbf{P}} + \bar{\mathbf{U}}^*,\tag{31}$$

where $\Pi_i = (\mathbf{A}_i', \Gamma_i')', \ \mathbf{V}_i = (\mathbf{v}_{i1}, \mathbf{v}_{i2}, ..., \mathbf{v}_{iT})'$

$$\mathbf{\bar{P}}_{(n+m)\times(n+k+1)} = \begin{pmatrix} \mathbf{I}_n & \mathbf{\bar{B}} \\ \mathbf{0} & \mathbf{\bar{C}} \end{pmatrix}, \ \mathbf{\bar{U}}^* = (\mathbf{0}, \mathbf{\bar{U}}), \tag{32}$$

Using Lemma 1 which forms the basis of all the theoretical results of this paper and assuming that the rank condition (9) is satisfied, it follows that

$$\frac{\mathbf{X}_{i}'\mathbf{\overline{M}F}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\overline{M}}_{g}\mathbf{F}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{33}$$

²Our analysis allows for cointegration in \mathbf{d}_{2t} as well. However, for the sake of simplicity we do not explicitly analyse this case which is a straightforward extension of the results presented in this paper.

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{g}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{34}$$

and

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\boldsymbol{\varepsilon}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{g}\boldsymbol{\varepsilon}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right). \tag{35}$$

Note that $\mathbf{F} \subset \mathbf{G}$ and hence that $T^{-1}\mathbf{X}_i'\mathbf{\bar{M}}_g\mathbf{F} = \mathbf{0}$. If the rank condition does not hold then again by Lemma 1 it follows that

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{F}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\mathbf{F}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{36}$$

$$\frac{\mathbf{X}_{i}'\mathbf{\bar{M}}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\bar{M}}_{q}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{37}$$

and

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\boldsymbol{\varepsilon}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\boldsymbol{\varepsilon}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right). \tag{38}$$

where

$$\bar{\mathbf{M}}_q = \mathbf{I}_T - \bar{\mathbf{Q}} (\bar{\mathbf{Q}}'\bar{\mathbf{Q}})^- \bar{\mathbf{Q}}', \text{ with } \bar{\mathbf{Q}} = \mathbf{G}\bar{\mathbf{P}}.$$
 (39)

Using the above results, noting that $T^{-1}\mathbf{X}_{i}'\mathbf{\bar{M}}_{q}\mathbf{X}_{i} = O_{p}(1)$, and assuming that the rank condition (9) is satisfied we have

$$\hat{\mathbf{b}}_{i} - \boldsymbol{\beta}_{i} = \left(\frac{\mathbf{X}_{i}'\mathbf{M}_{g}\mathbf{X}_{i}}{T}\right)^{-1} \left(\frac{\mathbf{X}_{i}'\mathbf{M}_{g}\boldsymbol{\varepsilon}_{i}}{T}\right) + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right). \tag{40}$$

Since ε_i is independently distributed of \mathbf{X}_i and $\mathbf{G} = (\mathbf{D}, \mathbf{F})$, then for a fixed T, $\lim_{N \to \infty} E\left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i\right) = \mathbf{0}$. The finite-T distribution of $\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i$ will be free of nuisance parameters as $N \to \infty$, but will depend on the probability density of ε_i . For N and T sufficiently large, the distribution of $\sqrt{T}\left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i\right)$ will be asymptotically normal if the rank condition (9) is satisfied and if N and T are of the same order of magnitudes, namely, if $T/N \to \kappa$ as N and $T \to \infty$, where κ is a positive finite constant. To see why this additional condition is needed, using (40) note that

$$\sqrt{T} \left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i \right) = \left(\frac{\mathbf{X}_i' \mathbf{M}_g \mathbf{X}_i}{T} \right)^{-1} \frac{\mathbf{X}_i' \mathbf{M}_g \boldsymbol{\varepsilon}_i}{\sqrt{T}} + O_p \left(\frac{\sqrt{T}}{N} \right) + O_p \left(\frac{1}{\sqrt{N}} \right), \tag{41}$$

and the asymptotic distribution of $\sqrt{T}\left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i\right)$ will be free of nuisance parameters only if $\sqrt{T}/N \to 0$, as $(N,T) \stackrel{j}{\to} \infty$. For this condition to be satisfied it is sufficient that $T/N \to \kappa$, as $(N,T) \stackrel{j}{\to} \infty$, where κ is a finite non-negative constant as N and $T \to \infty$.

In the case where there are no cointegrating relations amongst the elements of \mathbf{f}_t , and $\mathbf{d}_{3t} = \mathbf{0}$, the results simplify since by Lemma 2

$$\frac{\mathbf{X}_{i}'\mathbf{\bar{M}F}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\bar{M}}_{g}\mathbf{F}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right),\tag{42}$$

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{g}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right),\tag{43}$$

and

$$\frac{\mathbf{X}_{i}'\mathbf{\bar{M}}\varepsilon_{i}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\bar{M}}_{g}\varepsilon_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right). \tag{44}$$

Hence, the condition $\sqrt{T}/N \to 0$, as $(N,T) \xrightarrow{j} \infty$ will not be needed for the validity of the asymptotic distribution of $\sqrt{T} \left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i \right)$ for large N and T. The following theorem provides a formal statement of these results and the associated asymptotic distributions in the case where the rank condition (9) is satisfied.

Theorem 1 Consider the panel data model (1) and (2) and suppose that $\|\boldsymbol{\beta}_i\| < K$, $\|\boldsymbol{\Pi}_i\| < K$, Assumptions 1-5 hold. Let $\sqrt{T}/N \to 0$, as $(N,T) \xrightarrow{j} \infty$, and the rank condition (9) be satisfied. (a) - (N-asymptotic) The common correlated effects estimator, $\hat{\mathbf{b}}_i$, defined by (15) is unbiased for a fixed T > n + 2k + 1 and $N \to \infty$, in the sense that $\lim_{N\to\infty} E\left(\hat{\mathbf{b}}_i\right) = \boldsymbol{\beta}_i$. Under the additional assumption that $\varepsilon_{it} \sim IIDN(0, \sigma_i^2)$,

$$\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i \stackrel{d}{\to} N(\mathbf{0}, \boldsymbol{\Sigma}_{T, b_i}),$$
 (45)

as $N \to \infty$, where

$$\Sigma_{T,b_i} = T^{-1} \sigma_i^2 \Psi_{ig}^{-1}, \quad \Psi_{ig} = T^{-1} \left(\mathbf{X}_i' \mathbf{M}_g \mathbf{X}_i \right), \tag{46}$$

$$\mathbf{M}_q = \mathbf{I}_T - \mathbf{G}(\mathbf{G}'\mathbf{G})^{-1}\mathbf{G}',\tag{47}$$

and $\mathbf{G} = (\mathbf{g}_1, \mathbf{g}_2, ..., \mathbf{g}_T) = (\mathbf{F}, \mathbf{D})$. (b) - (Joint asymptotic) As $(N, T) \xrightarrow{j} \infty$ (in no particular order), $\hat{\mathbf{b}}_i$ is a consistent estimator of $\boldsymbol{\beta}_i$. Then, as $(N, T) \xrightarrow{j} \infty$

$$\sqrt{T}\left(\hat{\mathbf{b}}_i - \boldsymbol{\beta}_i\right) \stackrel{d}{\to} N(\mathbf{0}, \boldsymbol{\Sigma}_{b_i}),$$
 (48)

where

$$\Sigma_{b_i} = \sigma_i^2 \Sigma_i^{-1}. \tag{49}$$

An asymptotically unbiased estimator of Σ_{T,b_i} , as $N \to \infty$ for a fixed T > n + 2k + 1, is given by:

$$\hat{\Sigma}_{T,b_i} = \hat{\sigma}_i^2 \left(\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i \right)^{-1}, \tag{50}$$

where

$$\hat{\sigma}_i^2 = \frac{\left(\mathbf{y}_i - \mathbf{X}_i \hat{\mathbf{b}}_i\right)' \bar{\mathbf{M}} \left(\mathbf{y}_i - \mathbf{X}_i \hat{\mathbf{b}}_i\right)}{T - (n + m + k)}.$$
(51)

In the case where $(N,T) \xrightarrow{j} \infty$, a consistent estimator of Σ_{b_i} is given by

$$\hat{\Sigma}_{b_i} = \mathring{\sigma}_i^2 \left(\frac{\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i}{T} \right)^{-1}, \tag{52}$$

where

$$\mathring{\sigma}_{i}^{2} = \frac{\left(\mathbf{y}_{i} - \mathbf{X}_{i}\hat{\mathbf{b}}_{i}\right)' \bar{\mathbf{M}} \left(\mathbf{y}_{i} - \mathbf{X}_{i}\hat{\mathbf{b}}_{i}\right)}{T - (n + 2k + 1)}.$$
(53)

Remark 3 It is worth noting that despite the fact that under our Assumptions \mathbf{f}_t , y_{it} and \mathbf{x}_{it} are I(1) and cointegrated, in the results of Theorem 1 the rate of convergence of $\hat{\mathbf{b}}_i$ to $\boldsymbol{\beta}_i$ as $(N,T) \stackrel{j}{\to} \infty$ is \sqrt{T} and not T. It is helpful to develop some intuition behind this result. Since for N sufficiently large \mathbf{f}_t can be well approximated by the cross section averages we might as well consider the case where \mathbf{f}_t is observed. Abstracting from \mathbf{d}_t , and without loss of generality substitute (2) in (1) to obtain

$$y_{it} = \beta_i' (\Gamma_i' \mathbf{f}_t + \mathbf{v}_{it}) + \gamma_i' \mathbf{f}_t + \varepsilon_{it} = \vartheta_i' \mathbf{f}_t + \varkappa_{it}$$

where $\vartheta_i = \Gamma_i \beta_i + \gamma_i$ and $\varkappa_{it} = \varepsilon_{it} + \beta'_i v_{it}$. It is clear that if \mathbf{f}_t is I(1), as postulated under our assumptions, then for all values of β_i , \varkappa_{it} is I(0) and y_{it} and \mathbf{f}_t will be I(1) and cointegrated. Hence, it is easily seen that ϑ_i can be estimated superconsistently. However, the OLS estimator of β_i need not be superconsistent. To see this note that β_i can equivalently be estimated by regressing the residuals of y_{it} on \mathbf{f}_t on the residuals of \mathbf{x}_{it} on \mathbf{f}_t . Both these sets of residuals are stationary processes and the resulting estimator of β_i will be \sqrt{T} -consistent.

Remark 4 When the rank condition, (9), is not satisfied consistent estimation of the individual slope coefficients is not possible.

3.1 Pooled Estimators

We now examine the asymptotic properties of the pooled estimators. Focusing first on the MG estimator, and using (29) we have

$$\sqrt{N} \left(\hat{\mathbf{b}}_{MG} - \boldsymbol{\beta} \right) = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \boldsymbol{v}_{i} + \frac{1}{N} \sum_{i=1}^{N} \hat{\boldsymbol{\Psi}}_{iT}^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{F}}{T} \right) \boldsymbol{\gamma}_{i} + \frac{1}{N} \sum_{i=1}^{N} \hat{\boldsymbol{\Psi}}_{iT}^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \bar{\mathbf{M}} \boldsymbol{\varepsilon}_{i}}{T} \right), \tag{54}$$

In the case where the rank condition (9) is satisfied, again by Lemmas 1-3 we have

$$\frac{\sqrt{N}\left(\mathbf{X}_{i}'\mathbf{\overline{M}F}\right)}{T} = O_{p}\left(\frac{1}{\sqrt{T}}\right),$$

and

$$\frac{1}{N} \sum_{i=1}^{N} \hat{\mathbf{\Psi}}_{iT}^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \mathbf{\bar{M}} \varepsilon_{i}}{T} \right) = O_{p} \left(\frac{1}{\sqrt{T}} \right),$$

Thus

$$\sqrt{N}\left(\hat{\mathbf{b}}_{MG} - \boldsymbol{\beta}\right) = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \boldsymbol{v}_i + O_p\left(\frac{1}{\sqrt{T}}\right).$$

Hence

$$\sqrt{N}\left(\hat{\mathbf{b}}_{MG} - \boldsymbol{\beta}\right) \stackrel{d}{\to} N(\mathbf{0}, \boldsymbol{\Sigma}_{MG}), \text{ as } (N, T) \stackrel{j}{\to} \infty.$$
 (55)

The variance estimator for Σ_{MG} suggested by Pesaran (2006) is given by

$$\hat{\mathbf{\Sigma}}_{MG} = \frac{1}{N-1} \sum_{i=1}^{N} \left(\hat{\mathbf{b}}_{i} - \hat{\mathbf{b}}_{MG} \right) \left(\hat{\mathbf{b}}_{i} - \hat{\mathbf{b}}_{MG} \right)', \tag{56}$$

can be used. It can also be shown along identical lines to Pesaran (2006) that the mean group estimator is valid even if the rank condition is not satisfied. The following theorem summarises the results for the mean group estimator.

Theorem 2 Consider the panel data model (1) and (2) and suppose that Assumptions 1-5 hold. Then the Common Correlated Effects Mean Group estimator, \mathbf{b}_{MG} defined by (14), is asymptotically (for a fixed T and as $N \to \infty$) unbiased for $\boldsymbol{\beta}$, and as $(N,T) \xrightarrow{j} \infty$

$$\sqrt{N}\left(\hat{\mathbf{b}}_{MG} - \boldsymbol{\beta}\right) \stackrel{d}{\to} N(\mathbf{0}, \boldsymbol{\Sigma}_{MG}),$$

where Σ_{MG} is consistently estimated by (56).

This theorem does not require that the rank condition, (9), holds for any number, m, of unobserved factors so long as m is fixed, and does not impose any restrictions on the relative rates of expansion of N and T. But in the case where the rank condition is satisfied Assumption 3 can be relaxed and the factor loadings, γ_i , need not follow the random coefficient model. It would be sufficient that they are bounded.

Moving to the pooled estimator we have the following. $\hat{\mathbf{b}}_P$ defined by (17), can be written as

$$\sqrt{N}\left(\hat{\mathbf{b}}_{P} - \boldsymbol{\beta}\right) = \left(\frac{1}{N}\sum_{i=1}^{N} \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T}\right)^{-1} \left[\frac{1}{\sqrt{N}}\sum_{i=1}^{N} \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}(\mathbf{X}_{i}\boldsymbol{\upsilon}_{i} + \boldsymbol{\varepsilon}_{i})}{T} + \mathbf{q}_{NT}\right].$$
(57)

where

$$\mathbf{q}_{NT} = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \frac{\left(\mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{F}\right) \boldsymbol{\gamma}_{i}}{T}.$$
 (58)

Assuming random coefficients we note that $\gamma_i = \bar{\gamma} + \eta_i - \bar{\eta}$, where $\bar{\eta} = \frac{1}{N} \sum_{i=1}^N \eta_i$. Hence

$$\mathbf{q}_{NT} = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{F}}{T} \right) (\bar{\boldsymbol{\gamma}} - \bar{\boldsymbol{\eta}}) + \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{F}}{T} \right) \boldsymbol{\eta}_{i}.$$

Recall that when the rank condition is not satisfied $T^{-1}\left(\mathbf{X}_{i}^{\prime}\mathbf{\bar{M}F}\right)=O_{p}(1)$. But since $\mathbf{\bar{X}}^{\prime}\mathbf{\bar{M}}=\mathbf{0}$, we have

$$\sum_{i=1}^{N} \mathbf{X}_{i}' \mathbf{\bar{M}} \mathbf{F} \left(\mathbf{\bar{\gamma}} - \mathbf{\bar{\eta}} \right) = N \mathbf{\bar{X}}' \mathbf{\bar{M}} \mathbf{F} \left(\mathbf{\bar{\gamma}} - \mathbf{\bar{\eta}} \right) = \mathbf{0},$$

and it follows that

$$\mathbf{q}_{NT} = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \bar{\mathbf{M}} \mathbf{F}}{T} \right) \boldsymbol{\eta}_{i}$$

$$= \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{q} \mathbf{F}}{T} \right) \boldsymbol{\eta}_{i} + O_{p} \left(\frac{1}{\sqrt{N}} \right) + O_{p} \left(\frac{1}{\sqrt{T}} \right),$$
(59)

Substituting this result in (57), and making use of (34) and (35) we have

$$\sqrt{N} \left(\hat{\mathbf{b}}_{P} - \boldsymbol{\beta} \right) = \left(\frac{1}{N} \sum_{i=1}^{N} \frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{q} \mathbf{X}_{i}}{T} \right)^{-1} \left[\frac{1}{\sqrt{N}} \sum_{i=1}^{N} \frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{q} (\mathbf{X}_{i} \boldsymbol{v}_{i} + \boldsymbol{\varepsilon}_{i} + \mathbf{F} \boldsymbol{\eta}_{i})}{T} \right] + O_{p} \left(\frac{1}{\sqrt{T}} \right). \tag{60}$$

Hence, as $(N,T) \xrightarrow{j} \infty$

$$\sqrt{N}\left(\hat{\boldsymbol{b}} - \boldsymbol{\beta}\right) \stackrel{d}{\to} N(\boldsymbol{0}, \boldsymbol{\Sigma}_P^*),$$

where

$$\Sigma_P^* = \Psi^{*-1} \mathbf{R}^* \Psi^{*-1}, \tag{61}$$

$$\mathbf{\Psi}^* = \lim_{N \to \infty} \left(N^{-1} \sum_{i=1}^{N} \mathbf{\Sigma}_{iq} \right), \ \mathbf{R}^* = \lim_{N \to \infty} \left[N^{-1} \sum_{i=1}^{N} \left(\mathbf{\Sigma}_{iq} \mathbf{\Omega}_{v} \mathbf{\Sigma}_{iq} + \mathbf{Q}_{if} \mathbf{\Omega}_{\eta} \mathbf{Q}'_{if} \right) \right], \tag{62}$$

and Σ_{iq} and \mathbf{Q}_{if} are defined by

$$\Sigma_{iq} = p \lim_{T \to \infty} \left(T^{-1} \mathbf{X}_i' \bar{\mathbf{M}}_q \mathbf{X}_i \right) \text{ and } \mathbf{Q}_{if} = p \lim_{T \to \infty} \left(T^{-1} \mathbf{X}_i' \bar{\mathbf{M}}_q \mathbf{F} \right).$$
 (63)

The variance estimator for Σ_P^* suggested by Pesaran (2006) and given by

$$\hat{\Sigma}_P^* = \hat{\Psi}^{*-1} \hat{\mathbf{R}}^* \hat{\Psi}^{*-1}, \tag{64}$$

where

$$\hat{\mathbf{\Psi}}^* = N^{-1} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i}{T} \right), \tag{65}$$

$$\hat{\mathbf{R}}^* = \frac{1}{(N-1)} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i}{T} \right) \left(\hat{\mathbf{b}}_i - \hat{\mathbf{b}}_{MG} \right) \left(\hat{\mathbf{b}}_i - \hat{\mathbf{b}}_{MG} \right)' \left(\frac{\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i}{T} \right).$$
(66)

Again we summarise these results in the following theorem.

Theorem 3 Consider the panel data model (1) and (2) and suppose that Assumptions 1-5 hold. Then the Common Correlated Effects Pooled estimator, $\hat{\mathbf{b}}_P$, defined by (17) is asymptotically unbiased for $\boldsymbol{\beta}$, and as $(N,T) \xrightarrow{j} \infty$ we have

$$\sqrt{N}\left(\hat{\mathbf{b}}_P - \boldsymbol{\beta}\right) \stackrel{d}{\to} N(\mathbf{0}, \boldsymbol{\Sigma}_P^*),$$

where Σ_P^* is given by (61)-(63).

Overall we see that despite a number of differences in the above analysis, especially in terms of the results given in (33)-(35), compared to the results in Pesaran (2006), the conclusions are remarkably similar when the factors are assumed to follow unit root processes rather that weakly stationary ones.

4 Monte Carlo Design and Evidence

In this section we provide Monte Carlo evidence on the small sample properties of the CCEMG and the CCEP estimators. We also consider the two alternative principal component augmentation approaches discussed in Kapetanios and Pesaran (2006). The first PC approach applies the Bai and Ng (2002) procedure to $\mathbf{z}_{it} = (y_{it}, \mathbf{x}'_{it})'$ to obtain consistent estimates of the unobserved factors, and then uses the estimated factors to augment the regression (1), and thus produces consistent estimates of $\boldsymbol{\beta}$. We consider both pooled and mean group versions of this estimator which we refer to as PC1POOL and PC1MG. The second PC approach consists of extracting the principal component estimates of the unobserved factors from y_{it} and \mathbf{x}_{it} separately, regressing y_{it} and \mathbf{x}_{it} on their respective factor estimates separately, and then applying the standard pooled and mean group estimators, with no cross-sectional dependence adjustments, to the residuals of these regressions. We refer to the estimators based on this approach as PC2POOL and PC2MG, respectively.

The experimental design of the Monte Carlo study is closely related to the one used in Pesaran (2006). Consider the following data generating process (DGP):

$$y_{it} = \alpha_{i1}d_{1t} + \beta_{i1}x_{1it} + \beta_{i2}x_{2it} + \gamma_{i1}f_{1t} + \gamma_{i2}f_{2t} + \varepsilon_{it}, \tag{67}$$

and

$$x_{ijt} = a_{ij1}d_{1t} + a_{ij2}d_{2t} + \gamma_{ij1}f_{1t} + \gamma_{ij3}f_{3t} + v_{ijt}, j = 1, 2,$$
(68)

for i = 1, 2, ..., N, and t = 1, 2, ..., T. This DGP is a restricted version of the general linear model considered in Pesaran (2006), and sets n = k = 2, and m = 3, with $\alpha'_i = (\alpha_{i1}, 0)$, $\beta'_i = (\beta_{i1}, \beta_{i2})$, and $\gamma'_i = (\gamma_{i1}, \gamma_{i2}, 0)$, and

$$\mathbf{A}'_{i} = \begin{pmatrix} a_{i11} & a_{i12} \\ a_{i21} & a_{i22} \end{pmatrix}, \ \mathbf{\Gamma}'_{i} = \begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix}.$$

The observed common factors and the individual specific errors of \mathbf{x}_{it} are generated as independent stationary AR(1) processes with zero means and unit variances:

$$d_{1t} = 1$$
, $d_{2t} = \rho_d d_{2,t-1} + v_{dt}$, $t = -49, ...1, ..., T$,
 $v_{dt} \sim IIDN(0, 1 - \rho_d^2)$, $\rho_d = 0.5$, $d_{2,-50} = 0$,

$$\mathbf{v}_{ijt} = \rho_{vij} \mathbf{v}_{ijt-1} + v_{ijt}, \ t = -49, ...1, ..., T,$$

$$v_{ijt} \sim IIDN\left(0, 1 - \rho_{vij}^2\right), \ \mathbf{v}_{ji,-50} = 0,$$

and

$$\rho_{vij} \sim IIDU [0.05, 0.95], \text{ for } j = 1, 2,$$

but the unobserved common factors are generated as non-stationary processes:

$$f_{jt} = f_{jt-1} + v_{fj,t}$$
, for $j = 1, 2, 3$, $t = -49, ..., 0, ..., T$, $v_{fj,t} \sim IIDN(0,1)$, $f_{j,-50} = 0$, for $j = 1, 2, 3$.

The first 50 observations are discarded.

To illustrate the robustness of the CCE and PC estimators to the dynamics of the individual specific errors of y_{it} , these are generated as the (cross sectional) mixture of stationary heterogeneous AR(1) and MA(1) errors. Namely,

$$\varepsilon_{it} = \rho_{i\varepsilon}\varepsilon_{i,t-1} + \sigma_i\sqrt{1 - \rho_{i\varepsilon}^2}\omega_{it}, i = 1, 2, ..., N_1, t = -49, ..., 0, ..., T,$$

and

$$\varepsilon_{it} = \frac{\sigma_i}{\sqrt{1 + \theta_{i\varepsilon}^2}} \left(\omega_{it} + \theta_{i\varepsilon} \omega_{i,t-1} \right), i = N_1 + 1, ..., N, t = -49, ..., 0, ..., T,$$

where N_1 is the nearest integer of N/2,

$$\omega_{it} \sim IIDN\left(0,1\right), \ \sigma_{i}^{2} \sim IIDU\left[0.5,1.5\right], \ \rho_{i\varepsilon} \sim IIDU\left[0.05,0.95\right], \ \theta_{i\varepsilon} \sim IIDU\left[0,1\right].$$

 ρ_{vij} , $\rho_{i\varepsilon}$, $\theta_{i\varepsilon}$ and σ_i are not changed across replications. The first 49 observations are discarded. The factor loadings of the observed common effects, α_{i1} , and $vec(\mathbf{A}_i) = (a_{i11}, a_{i21}, a_{i12}, a_{i22})'$ are generated as IIDN(1,1), and $IIDN(0.5\tau_4, 0.5 \mathbf{I}_4)$, where $\tau_4 = (1,1,1,1)'$, and are not changed across replications. They are treated as fixed effects. The parameters of the unobserved common effects in the \mathbf{x}_{it} equation are generated independently across replications as

$$\Gamma'_{i} = \begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix} \sim IID \begin{pmatrix} N(0.5, 0.50) & 0 & N(0, 0.50) \\ N(0, 0.50) & 0 & N(0.5, 0.50) \end{pmatrix}.$$

For the parameters of the unobserved common effects in the y_{it} equation, γ_i , we considered two different sets that we denote by \mathcal{A} and \mathcal{B} . Under set \mathcal{A} , γ_i are drawn such that the rank condition is satisfied, namely

$$\gamma_{i1} \sim IIDN(1, 0.2), \ \gamma_{i2A} \sim IIDN(1, 0.2), \ \gamma_{i3} = 0,$$

and

$$E\left(\tilde{\mathbf{\Gamma}}_{i\mathcal{A}}\right) = \left(E\left(\boldsymbol{\gamma}_{i\mathcal{A}}\right), E\left(\mathbf{\Gamma}_{i}\right)\right) = \begin{pmatrix} 1 & 0.5 & 0 \\ 1 & 0 & 0 \\ 0 & 0 & 0.5 \end{pmatrix}.$$

Under set \mathcal{B}

$$\gamma_{i1} \sim IIDN(1, 0.2), \ \gamma_{i2B} \sim IIDN(0, 1), \gamma_{i3} = 0,$$

so that

$$E\left(\tilde{\mathbf{\Gamma}}_{i\mathcal{B}}\right) = \left(E\left(\boldsymbol{\gamma}_{i\mathcal{B}}\right), E\left(\mathbf{\Gamma}_{i}\right)\right) = \begin{pmatrix} 1 & 0.5 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0.5 \end{pmatrix},$$

and the rank condition is not satisfied. For each set we conducted two different experiments:

- Experiment 1 examines the case of heterogeneous slopes with $\beta_{ij} = 1 + \eta_{ij}$, j = 1, 2, and $\eta_{ij} \sim IIDN(0, 0.04)$, across replications.
- Experiment 2 considers the case of homogeneous slopes with $\beta_i = \beta = (1, 1)'$.

The two versions of experiment 1 will be denoted by $1\mathcal{A}$ and $1\mathcal{B}$, and those of experiment 2 by $2\mathcal{A}$ and $2\mathcal{B}$. For this Monte Carlo study we also computed the CCEMG and the CCEP estimators as well as the associated "infeasible" estimators (MG and Pooled) that include f_{1t} and f_{2t} in the regressions of y_{it} on $(d_{1t}, \mathbf{x}_{it})$, and the "naive" estimators that excludes these factors. The naive estimators illustrate the extent of bias and size distortions that can occur if the error cross section dependence is ignored.

In relation to the infeasible pooled estimator, it is important to note that this estimator although unbiased under all the four sets of experiments, it need not be efficient since in these experiments the slope coefficients, β_i , and/or error variances, σ_i^2 , differ across i. As a result the CCE or PC augmented estimators may in fact dominate the infeasible estimator in terms of RMSE, particularly in the case of experiments $1\mathcal{A}$ and $1\mathcal{B}$ where the slopes as well as the error variances are allowed to vary across i.

Another important consideration worth bearing in mind when comparing the CCE and the PC type estimators is the fact that the computation of the PC augmented estimators assumes that m = 3, the number of unobserved factors, is known. In practice, m might be difficult to estimate accurately particularly when N or T happen to be smaller than 50.

By contrast the CCE type estimators are valid for any fixed m and do not require a prior estimate for m.

Each experiment was replicated 2000 times for the (N, T) pairs with N, T = 20, 30, 50, 100, 200. In what follows we shall focus on β_1 (the cross section mean of β_{i1}). Results for β_2 are very similar and will not be reported. Finally, for completeness we state below the exact formulae for the variance estimators used for the different estimators. The non-parametric variance estimators of the mean group estimators, $\tilde{\mathbf{b}}_{MG} = N^{-1} \sum_{i=1}^{N} \tilde{\mathbf{b}}_{i}$, are computed as

$$\widehat{Var}(\widetilde{\mathbf{b}}_{MG}) = \frac{1}{N(N-1)} \sum_{i=1}^{N} \left(\widetilde{\mathbf{b}}_{i} - \widetilde{\mathbf{b}}_{MG} \right) \left(\widetilde{\mathbf{b}}_{i} - \widetilde{\mathbf{b}}_{MG} \right)', \tag{69}$$

where

$$egin{aligned} ilde{\mathbf{b}}_i &= \left(\mathbf{X}_i' ilde{\mathbf{M}}_x \mathbf{X}_i
ight)^{-1} \mathbf{X}_i' ilde{\mathbf{M}}_x ilde{\mathbf{M}}_y \mathbf{y}_i, \ ilde{\mathbf{M}}_x &= \mathbf{I}_T - ilde{\mathbf{H}}_x \left(ilde{\mathbf{H}}_x' ilde{\mathbf{H}}_x
ight)^{-} ilde{\mathbf{H}}_x', \ ilde{\mathbf{M}}_y &= \mathbf{I}_T - ilde{\mathbf{H}}_y \left(ilde{\mathbf{H}}_y' ilde{\mathbf{H}}_y
ight)^{-} ilde{\mathbf{H}}_y'. \end{aligned}$$

For the CCEMG estimator, $\tilde{\mathbf{H}}_x = \tilde{\mathbf{H}}_y = \bar{\mathbf{H}} = (\mathbf{D}, \bar{\mathbf{Z}})$, so that $\tilde{\mathbf{b}}_i = \hat{\mathbf{b}}_i$, which is defined by (15); for the PC1MG estimator, $\tilde{\mathbf{H}}_x = \tilde{\mathbf{H}}_y = \hat{\mathbf{F}}_z$, where $\hat{\mathbf{F}}_z$ is a $T \times (n+m)$ matrix of extracted factors from \mathbf{Z}_i for all i, together with observed common factors; for the PC2MG estimator $\tilde{\mathbf{H}}_x = \hat{\mathbf{F}}_x$ and $\tilde{\mathbf{H}}_y = \hat{\mathbf{F}}_y$, where $\hat{\mathbf{F}}_x$ and $\hat{\mathbf{F}}_y$ are $T \times (n_x + m_x)$ and $T \times (n_y + m_y)$ matrices of extracted factors from \mathbf{X}_i and \mathbf{y}_i respectively for all i, together with the observed common factors with n_x and n_y being the number of observed common factors in \mathbf{X}_i and \mathbf{y}_i respectively, and m_x and m_y defined similarly; for the infeasible mean group estimator, $\tilde{\mathbf{H}}_x = \tilde{\mathbf{H}}_y = \mathbf{F}_y$, which is a $T \times m_y$ matrix of unobserved factors in \mathbf{y}_i ; for the naive mean group estimator, $\tilde{\mathbf{H}}_x = \tilde{\mathbf{H}}_y = \mathbf{D}$. Next, the non-parametric variance of the pooled estimator, $\tilde{\mathbf{b}}_P$, is computed as

$$\widehat{Var}(\tilde{\mathbf{b}}_P) = N^{-1} \tilde{\mathbf{\Psi}}^{-1} \tilde{\mathbf{R}} \tilde{\mathbf{\Psi}}^{-1}, \tag{70}$$

where

$$\begin{split} \tilde{\mathbf{b}}_P &= \left(\sum_{i=1}^N \mathbf{X}_i' \tilde{\mathbf{M}}_x \mathbf{X}_i\right)^{-1} \sum_{i=1}^N \mathbf{X}_i' \tilde{\mathbf{M}}_x \tilde{\mathbf{M}}_y \mathbf{y}_i, \\ \tilde{\mathbf{\Psi}} &= N^{-1} \sum_{i=1}^N \left(\frac{\mathbf{X}_i' \tilde{\mathbf{M}}_x \mathbf{X}_i}{T}\right), \\ \tilde{\mathbf{R}} &= \frac{1}{(N-1)} \sum_{i=1}^N \left(\frac{\mathbf{X}_i' \tilde{\mathbf{M}}_x \mathbf{X}_i}{T}\right) \left(\tilde{\mathbf{b}}_i - \tilde{\mathbf{b}}_{MG}\right) \left(\tilde{\mathbf{b}}_i - \tilde{\mathbf{b}}_{MG}\right)' \left(\frac{\mathbf{X}_i' \tilde{\mathbf{M}}_x \mathbf{X}_i}{T}\right). \end{split}$$

4.1 Results

Results of experiments 1A, 2A, 1B, 2B are summarized in Tables 1 to 4, respectively. We also provide results for the naive estimator (that excludes the unobserved factors or their

estimates) and the infeasible estimator (that includes the unobserved factors as additional regressors) for comparison purposes. But for the sake of brevity we include the simulation results for these estimators only for experiment $1\mathcal{A}$ reported in Table 1.

As can be seen from Table 1 the naive estimator is substantially biased, performs very poorly and is subject to large size distortions; an outcome that continues to apply in the case of other experiments (not reported here). In contrast, the feasible CCE estimators perform well, have bias that are close to the bias of the infeasible estimators, show little size distortions even for relatively small values of N and T, and their RMSE falls steadily with increases in N and/or T. These results are quite similar to the results presented in Pesaran (2006), and illustrate the robustness of the CCE estimators to the presence of unit roots in the unobserved common factors. This is important since it obviates the need for pre-testing involving unobserved factors.

The CCE estimators perform well, in both heterogeneous and homogeneous cases, and irrespective of whether the rank condition is satisfied, although the CCE estimators with rank deficiency have sightly higher RMSEs than those with full rank. The RMSEs of the CCE estimators of Tables 1 and 3 (heterogeneous case) are higher than those reported in Tables 2 and 4 for the homogeneous case. The sizes of the t-test based on the CCE estimators are very close to the nominal 5% level. In the case of full rank, the power of the tests for the CCE estimators are much higher than in the rank deficient case. Finally, not surprisingly the power of the tests for the CCE estimators in the homogeneous case is higher than that in the heterogeneous case.

It is also important to note that the small sample properties of the CCE estimator does not seem to be much affected by the residual serial correlation of the idiosyncratic errors, ε_{it} . The robustness of the CCE estimator to the short run dynamics is particularly helpful in practice where typically little is known about such dynamics. In fact a comparison of the results for the CCEP estimator with the infeasible counterpart given in Table 1 shows that the former can even be more efficient (in the RMSE sense). For example the RMSE of the CCEP for N = T = 50 is 3.97 whilst the RMSE of the infeasible pooled estimator is 4.31. This might seem counter intuitive at first, but as indicated above the infeasible estimator does not take account of the residual serial correlation of the idiosyncratic errors, but the CCE estimator does allow for such possibilities indirectly through the use of the cross section averages that partly embody the serial correlation properties of \mathbf{f}_t and ε_{it} 's.

Consider now the PC augmented estimators and recall that they are computed assuming the true number of common factors is known. The results summarized in Tables 1-4 bear some resemblance to those presented in Kapetanios and Pesaran (2006). The bias and RMSEs of the PC1POOL and PC1MG estimators improve as both N and T increase, but

the t-tests based on these estimators substantially over-reject the null hypothesis. The PC2POOL and PC2MG estimators perform even worse. The biases of the PC estimators are always larger in absolute value than the respective biases of the CCE estimators. The size distortion of the PC augmented estimators is particularly pronounced in the case of the experiments $1\mathcal{A}$ and $2\mathcal{A}$ (in Tables 1 and 2) where the full rank conditions are met. It is also interesting that in the case of some of the experiments the bias distortions of the tests based on the PC augmented estimators do not improve even for relatively large N and T. An interesting distinction arises when comparing results for experiments $1\mathcal{A}$ and $1\mathcal{B}$. For $1\mathcal{A}$ (heterogeneous slopes and full rank) results are very poor for small values of N and T but improve considerably as N rises and less perceptibly as T rises. For experiment $1\mathcal{B}$ (heterogeneous slopes and rank deficient) results are much better for small values of N and T. Finally, it is worth noting that in both cases the performance of the PC estimators actually get worse when N is small and kept small but T rises. This may be related to the fact that the accuracy of the factor estimates depends on the minimum of N and T.

5 Conclusions

Recently, there has been increased focus in the panel data literature on problems arising in estimation and inference when the standard assumption that the errors of the panel regression are cross-sectionally uncorrelated, is violated. When the errors of a panel regression are cross-sectionally correlated then standard estimation methods do not necessarily produce consistent estimates of the parameters of interest. An influential strand of the relevant literature provides a convenient parametrisation of the problem in terms of a factor model for the error terms.

Pesaran (2006) adopts an error multifactor structure and suggests new estimators that take into account cross-sectional dependence, making use of cross-sectional averages of the dependent and explanatory variables. However, he focusses on the case of weakly stationary factors that could be restrictive in some applications. This paper provides a formal extension of the results of Pesaran (2006) to the case where the unobserved factors are allowed to follow unit root processes. It is shown that the main results of Pesaran continue to hold in this more general case. This is certainly of interest given the fact that usually there is a large difference between results obtained for unit root and stationary processes. When we consider the small sample properties of the new estimators, we observe that again the results accord with the conclusions reached in the stationary case, lending further support to the use of the CCE estimators irrespective of the order of integration of the data observed.

References

- Anselin, L. (2001): "Spatial Econometrics," in A Companion to Theoretical Econometrics, ed. by B. Baltagi. Blackwell, Oxford.
- BAI, J. (2003): "Inferential Theory for Factor Models of Large Dinensions," *Econometrica*, 71, 135–173.
- Bai, J., and S. Ng (2002): "Determining the Number of Factors in Approximate Factor Models," *Econometrica*, 70, 191–221.
- Chamberlain, G., and M. Rothschild (1983): "Arbitrage, Factor Structure and Mean-Variance Analysis in Large Asset Markets," *Econometrica*, 51, 1305–1324.
- COAKLEY, J., A. FUERTES, AND R. P. SMITH (2002): "A Principal Components Approach to Cross-Section Dependence in Panels," *Mimeo, Birkbeck College, University of London*.
- Conley, T. G., and B. Dupor (2003): "A Spatial Analysis of Sectoral Complementarity," Journal of Political Economy, 111, 311–352.
- Conley, T. G., and G. Topa (2002): "Socio-economic Distance and Spatial Patterns in Unemployment," *Journal of Applied Econometrics*, 17, 303–327.
- Connor, G., and R. Korajzcyk (1986): "Performance Measurement with the Arbitrage Pricing Theory: A New Framework for Analysis," *Journal of Financial Economics*, 15, 373–394.
- ———— (1988): "Risk and Return in an Equilibrium APT: Application to a new test methodology," *Journal of Financial Economics*, 21, 255–289.
- DAVIDSON, J., AND R. DE JONG (2000): "The Functional Central Limit Theorem and Weak Convergence to Stochastic Integrals I: Weakly Dependent Processes," *Econometric Theory*, 16, 621–642.
- FORNI, M., M. HALLIN, M. LIPPI, AND L. REICHLIN (2000): "The Generalised Factor Model: Identification and Estimation," *Review of Economics and Statistics*, 82, 540–554.
- FORNI, M., AND L. REICHLIN (1998): "Let's Get Real: A Factor Analytical Approach to Disaggregated Business Cycle Dynamics," *Review of Economic Studies*, 65, 453–473.
- Kapetanios, G., and M. H. Pesaran (2006): "Alternative Approaches to Estimation and Inference in Large Multifactor Panels: Small Sample Results with an Application to

- Modelling of Asset Returns," in *The Refinement of Econometric Estimation and Test Procedures: Finite Sample and Asymptotic Analysis*, ed. by G. D. A. Phillips, and E. Tzavalis. Cambridge University Press, Cambridge.
- LEE, K. C., AND M. H. PESARAN (1993): "The Role of Sectoral Interactions in Wage Determination in the UK Economy," *The Economic Journal*, 103, 21–55.
- PESARAN, M. H. (2006): "Estimation and Inference in Large Heterogeneous panels with a Multifactor Error Structure," *Econometrica*, 74, 967–1012.
- Pesaran, M. H., T. Schuermann, and S. M. Weiner (2004): "Modeling Regional Interdependencies using a Global Error-Correcting Macroeconomic Model," *Journal of Business Economics and Statistics, (with Discussions and a Rejoinder)*, 22, 129–181.
- PESARAN, M. H., AND R. P. SMITH (1995): "Estimating Long-Run Relationships from Dynamic Heterogeneous Panels," *Journal of Econometrics*, 68, 79–113.
- STOCK, J. H., AND M. W. WATSON (2002): "Macroeconomic Forecasting Using Diffusion Indices," *Journal of Business and Economic Statistics*, 20, 147–162.
- Stone, R. (1947): "On the Interdependence of Blocks of Transactions," Supplement of the Journal of the Royal Statistical Society, 9, 1–45.

6 Appendix

Lemma 1 Under Assumptions 1-5 and if $\sum_{\ell=0}^{\infty} \mathbf{W}_{\ell}$ is of reduced rank where \mathbf{W}_{ℓ} , $\ell=0,...$, is defined in Remark 2, then as $(N,T) \xrightarrow{j} \infty$,

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{71}$$

$$\frac{\mathbf{X}_{i}'\mathbf{\overline{M}F}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\overline{M}}_{q}\mathbf{F}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right),\tag{72}$$

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\varepsilon_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\varepsilon_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right) + O_{p}\left(\frac{1}{N}\right). \tag{73}$$

Proof. We start by proving (71). The proof for (72) and (73) follow similarly. We need to determine the order of probability of $\frac{\mathbf{X}_t'\bar{\mathbf{M}}\mathbf{X}_i}{T} - \frac{\mathbf{X}_t'\bar{\mathbf{M}}_q\mathbf{X}_i}{T}$. We split the problem in two parts. By remark 2, \mathbf{f}_t may be represented, via a common trends representation, by a set of r stationary components and m-r random walk components. Without loss of generality we disregard the deterministic component \mathbf{d}_{1t} in the analysis. Let \mathbf{K}_1 denote the matrix of observations on the nonstationary components of the common trends representation of \mathbf{f}_t and the nonstationary components of \mathbf{d}_t . Let \mathbf{K}_2 denote the matrix of observations on the stationary components of the common trends representation of \mathbf{f}_t and the stationary components of \mathbf{d}_t . Note that the transformations needed to get $(\mathbf{K}_1, \mathbf{K}_2)$ from \mathbf{G} simply involve nonsingular rotations of \mathbf{f}_t . Let

$$egin{aligned} ar{\mathbf{H}}_j &= \mathbf{K}_j ar{\mathbf{P}}_j + ar{\mathbf{U}}^*, \ ar{\mathbf{M}}_{jq} &= \mathbf{I}_T - ar{\mathbf{Q}}_j \left(ar{\mathbf{Q}}_j' ar{\mathbf{Q}}_j
ight)^- ar{\mathbf{Q}}_j', \ ar{\mathbf{M}}_j &= \mathbf{I}_T - ar{\mathbf{H}}_j \left(ar{\mathbf{H}}_i' ar{\mathbf{H}}_j
ight)^- ar{\mathbf{H}}_j, \end{aligned}$$

with $\bar{\mathbf{Q}}_j = \mathbf{K}_j \bar{\mathbf{P}}_j$ for j = 1, 2 where $\bar{\mathbf{P}}_j$ is a nonsingular transformation of a partition of $\bar{\mathbf{P}}$ comformable to $(\mathbf{K}_1, \mathbf{K}_2)$. Then, the order of probability of $\frac{\mathbf{X}_i' \bar{\mathbf{M}} \mathbf{X}_i}{T} - \frac{\mathbf{X}_i' \bar{\mathbf{M}}_q \mathbf{X}_i}{T}$ is equal to the maximum of the orders of probability of $\frac{\mathbf{X}_i' \bar{\mathbf{M}}_1 \mathbf{X}_i}{T} - \frac{\mathbf{X}_i' \bar{\mathbf{M}}_1 q \mathbf{X}_i}{T}$ and $\frac{\tilde{\mathbf{X}}_i' \bar{\mathbf{M}}_2 \tilde{\mathbf{X}}_i}{T} - \frac{\tilde{\mathbf{X}}_i' \bar{\mathbf{M}}_2 q \tilde{\mathbf{X}}_i}{T}$ where $\tilde{\mathbf{X}}_i$ are the residuals of \mathbf{X}_i when regressed on \mathbf{K}_1 . To see this note that

$$\frac{\mathbf{X}_i'\mathbf{\bar{M}}\mathbf{X}_i}{T} = \frac{\mathbf{\tilde{X}}_i'\mathbf{\tilde{M}}_2\mathbf{\tilde{X}}_i}{T},$$

and

$$rac{\mathbf{X}_i' \mathbf{ ilde{M}}_q \mathbf{X}_i}{T} = rac{\mathbf{ ilde{X}}_{qi}' \mathbf{ ilde{M}}_{2q} \mathbf{ ilde{X}}_{qi}}{T},$$

where $\tilde{\mathbf{X}}_i = \bar{\mathbf{M}}_1 \mathbf{X}_i$, $\tilde{\mathbf{X}}_{qi} = \bar{\mathbf{M}}_{1q} \mathbf{X}_i$,

$$\mathbf{\tilde{M}}_2 = \mathbf{I}_T - \mathbf{\bar{M}}_1 \mathbf{\bar{H}}_2 \left(\mathbf{\bar{H}}_2' \mathbf{\bar{M}}_1 \mathbf{\bar{H}}_2 \right)^- \mathbf{\bar{H}}_2' \mathbf{\bar{M}}_1,$$

and

$$\mathbf{ ilde{M}}_{2q} = \mathbf{I}_T - \mathbf{ar{M}}_{1q} \mathbf{ar{Q}}_2 \left(\mathbf{ar{Q}}_2' \mathbf{ar{M}}_{1q} \mathbf{ar{Q}}_2
ight)^- \mathbf{ar{Q}}_2' \mathbf{ar{M}}_{1q}.$$

Then

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T} - \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\mathbf{X}_{i}}{T} \leq C\left(\left\|\frac{\mathbf{X}_{i}'\left(\bar{\mathbf{M}}_{1} - \bar{\mathbf{M}}_{1q}\right)\tilde{\mathbf{M}}_{2}\tilde{\mathbf{X}}_{i}}{T}\right\| + \left\|\frac{\tilde{\mathbf{X}}_{i}'\left(\tilde{\mathbf{M}}_{2} - \tilde{\mathbf{M}}_{2q}\right)\tilde{\mathbf{X}}_{i}}{T}\right\|\right),$$

for some positive constant, C. The desired result then follows easily from Lemma 2, Lemmas A.2 and A.3 of Pesaran (2006). \blacksquare

Lemma 2 Under Assumptions 1-5 and if $\sum_{\ell=0}^{\infty} \mathbf{W}_{\ell}$ is of full rank where \mathbf{W}_{ℓ} , $\ell=0,...$, is defined in Remark 2, then as $(N,T) \xrightarrow{j} \infty$,

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right),\tag{74}$$

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}\mathbf{F}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{q}\mathbf{F}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right),\tag{75}$$

$$\frac{\mathbf{X}_{i}'\mathbf{\bar{M}}\varepsilon_{i}}{T} = \frac{\mathbf{X}_{i}'\mathbf{\bar{M}}_{q}\varepsilon_{i}}{T} + O_{p}\left(\frac{1}{\sqrt{NT}}\right). \tag{76}$$

Proof. For simplicity but without loss of generality we assume that \mathbf{d}_{3t} is empty. We start by proving (74). Throughout the proof C_i , i = 1, ..., denote different positive constants. We need to determine the order of probability of $\frac{\mathbf{X}_i'\bar{\mathbf{M}}\mathbf{X}_i}{T} - \frac{\mathbf{X}_i'\bar{\mathbf{M}}_q\mathbf{X}_i}{T}$. But this is equal to

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}\left(\bar{\mathbf{H}}'\bar{\mathbf{H}}\right)^{-}\bar{\mathbf{H}}'\mathbf{X}_{i}}{T} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}\left(\bar{\mathbf{Q}}'\bar{\mathbf{Q}}\right)^{-}\bar{\mathbf{Q}}'\mathbf{X}_{i}}{T} =$$

$$\left(\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{H}}'\mathbf{X}_{i}}{T^{3/2}}\right) - \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{Q}}'\mathbf{X}_{i}}{T^{3/2}}\right) \leq$$

$$C_{1}\left\|\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right\| + C_{2}\left\|\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}} - \frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right\|.$$
(77)

We examine the first term of (77) first. We have

$$\frac{\mathbf{X}_i'\bar{\mathbf{H}}}{T^{3/2}} - \frac{\mathbf{X}_i'\bar{\mathbf{Q}}}{T^{3/2}} = \frac{\mathbf{X}_i'\bar{\mathbf{U}}}{T^{3/2}},$$

where $\bar{\mathbf{U}} = \frac{1}{N} \sum_{i=1}^{N} \mathbf{U}_i$. But

$$\frac{\mathbf{X}_i'\bar{\mathbf{U}}}{T^{3/2}} \le C_3 \left\| \frac{\mathbf{F}'\bar{\mathbf{U}}}{T^{3/2}} + \frac{\mathbf{D}'\bar{\mathbf{U}}}{T^{3/2}} + \frac{\mathbf{V}_i'\bar{\mathbf{U}}}{T^{3/2}} \right\|,\tag{78}$$

and

$$\frac{\mathbf{V}_{i}'\bar{\mathbf{U}}}{T^{3/2}} = O_{p}\left(\frac{1}{T\sqrt{N}}\right) + O_{p}\left(\frac{1}{N\sqrt{T}}\right),\tag{79}$$

by Lemma A.2 of Pesaran (2006). A similar treatment can be applied to the first and second terms of the RHS of (78). We therefore examine the first term only. Consider the ℓ^{th} row of $T^{-3/2}\left(\mathbf{F}'\bar{\mathbf{U}}\right)$ and note that it can be written as $T^{-3/2}\left(\sum_{t=1}^T f_{\ell t}\bar{\mathbf{u}}_t'\right)$. Consider now the limit of $T^{-3/2}\sum_{t=1}^T f_{\ell t}\bar{\mathbf{u}}_t$. First note that Assumptions 1-3 and Remark 1, via Theorem 4.1 of Davidson and De Jong (2000), guarantee that \mathbf{u}_{it} can play the role of the integrator in a functional central limit theorem, i.e. the limit of $T^{-1}\sum_{t=1}^T f_{\ell t}\mathbf{u}_{it}'$ is a stochastic integral and therefore $O_p(1)$. Further note that for all finite N as well as when $N \to \infty$, the limit of $T^{-1}\sum_{t=1}^T f_{\ell t}\sqrt{N}\bar{\mathbf{u}}_t$ is a stochastic integral since assumption 3 and Remark 1 state that \mathbf{u}_{it}' is a $L_2 - NED$ process of size 1/2. This assumption, in turn, implies that $\bar{\mathbf{u}}_t$ is a $L_2 - NED$ process of size 1/2 for all finite N and as $N \to \infty$, and the result holds again via Theorem 4.1 of Davidson and De Jong (2000). To see this we need to show that for some positive integer s > 0

$$E\left\{\frac{1}{\sqrt{N}}\sum_{i=1}^{N}\mathbf{u}_{it} - E\left(\frac{1}{\sqrt{N}}\sum_{i=1}^{N}\mathbf{u}_{it} \middle| t - s\right)\right\}^{2} = O(s^{-1}).$$
 (80)

But

$$E\left\{\frac{1}{\sqrt{N}}\sum_{i=1}^{N}\mathbf{u}_{it} - E\left(\frac{1}{\sqrt{N}}\sum_{i=1}^{N}\mathbf{u}_{it} \middle| t - s\right)\right\}^{2} \le \frac{1}{N}\sum_{i=1}^{N}E\left[\mathbf{u}_{it} - E\left(\mathbf{u}_{it}\middle| t - s\right)\right]^{2}, \quad (81)$$

and since $E\left[\mathbf{u}_{it} - E\left(\mathbf{u}_{it}|t-s\right)\right]^2 = O(s^{-1})$ for all i, the desired result follows. From the above we also have

$$\frac{\mathbf{F}'\bar{\mathbf{U}}}{T^{3/2}} = O_p\left(\frac{1}{\sqrt{NT}}\right). \tag{82}$$

Similarly,

$$\frac{\mathbf{D}'\bar{\mathbf{U}}}{T^{3/2}} = O_p\left(\frac{1}{\sqrt{NT}}\right),\tag{83}$$

and hence,

$$\frac{\mathbf{X}_i'\bar{\mathbf{U}}}{T^{3/2}} = O_p\left(\frac{1}{\sqrt{NT}}\right). \tag{84}$$

We next examine the second term of (77). We have

$$\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^2} - \frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^2} \le C_4 \left\| \frac{\bar{\mathbf{Q}}'\bar{\mathbf{U}}}{T^2} + \frac{\bar{\mathbf{U}}'\bar{\mathbf{U}}}{T^2} \right\|. \tag{85}$$

By Lemma A.2 of Pesaran (2006) it follows that

$$\frac{\bar{\mathbf{U}}'\bar{\mathbf{U}}}{T^2} = O_p\left(\frac{1}{NT}\right). \tag{86}$$

Further,

$$\frac{\bar{\mathbf{Q}}'\bar{\mathbf{U}}}{T^2} = O_p\left(\frac{\mathbf{F}'\bar{\mathbf{U}}}{T^2}\right) + O_p\left(\frac{\mathbf{D}'\bar{\mathbf{U}}}{T^2}\right) = O_p\left(\frac{1}{T\sqrt{N}}\right),\tag{87}$$

by (82). Thus,

$$\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^2} - \frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^2} = O_p\left(\frac{1}{T\sqrt{N}}\right). \tag{88}$$

We now consider (75). We have

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}\left(\bar{\mathbf{H}}'\bar{\mathbf{H}}\right)^{-}\bar{\mathbf{H}}'\mathbf{F}}{T} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}\left(\bar{\mathbf{Q}}'\bar{\mathbf{Q}}\right)^{-}\bar{\mathbf{Q}}'\mathbf{F}}{T} =$$

$$\left(\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{H}}'\mathbf{F}}{T^{3/2}}\right) - \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{Q}}'\mathbf{F}}{T^{3/2}}\right) \leq$$

$$C_{5} \left\|\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right\| + C_{6} \left\|\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}} - \frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right\| + C_{7} \left\|\frac{\bar{\mathbf{H}}'\mathbf{F}}{T^{3/2}} - \frac{\bar{\mathbf{Q}}'\mathbf{F}}{T^{3/2}}\right\|.$$
(89)

We only need to examine the third term of the RHS of (89). But

$$\frac{\bar{\mathbf{H}}'\mathbf{F}}{T^{3/2}} - \frac{\bar{\mathbf{Q}}'\mathbf{F}}{T^{3/2}} = \frac{\bar{\mathbf{U}}'\mathbf{F}}{T^{3/2}}.$$

But then by (82) and (84), (75) follows.

Finally, we consider (76). The treatment follows closely that in (77) and (89). In particular

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}\left(\bar{\mathbf{H}}'\bar{\mathbf{H}}\right)^{-}\bar{\mathbf{H}}'\varepsilon_{i}}{T} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}\left(\bar{\mathbf{Q}}'\bar{\mathbf{Q}}\right)^{-}\bar{\mathbf{Q}}'\varepsilon_{i}}{T} = \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{H}}'\varepsilon_{i}}{T^{3/2}}\right) - \left(\frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right)\left(\frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right)^{-}\left(\frac{\bar{\mathbf{Q}}'\varepsilon_{i}}{T^{3/2}}\right) \leq C_{8} \left\|\frac{\mathbf{X}_{i}'\bar{\mathbf{H}}}{T^{3/2}} - \frac{\mathbf{X}_{i}'\bar{\mathbf{Q}}}{T^{3/2}}\right\| + C_{9} \left\|\frac{\bar{\mathbf{H}}'\bar{\mathbf{H}}}{T^{2}} - \frac{\bar{\mathbf{Q}}'\bar{\mathbf{Q}}}{T^{2}}\right\| + C_{10} \left\|\frac{\bar{\mathbf{H}}'\varepsilon_{i}}{T^{3/2}} - \frac{\bar{\mathbf{Q}}'\varepsilon_{i}}{T^{3/2}}\right\|. \tag{90}$$

But,

$$\frac{\bar{\mathbf{H}}'\boldsymbol{\varepsilon}_i}{T^{3/2}} - \frac{\bar{\mathbf{Q}}'\boldsymbol{\varepsilon}_i}{T^{3/2}} = \frac{\bar{\mathbf{U}}'\boldsymbol{\varepsilon}_i}{T^{3/2}}.$$

By Lemma A.2 of Pesaran (2006), it follows that

$$\frac{\bar{\mathbf{U}}'\boldsymbol{\varepsilon}_i}{T^{3/2}} = O_p\left(\frac{1}{T\sqrt{N}}\right) + O_p\left(\frac{1}{N\sqrt{T}}\right).$$

Hence, (76) follows. \blacksquare

Table 1: Small Sample Properties of Common Correlated Effects Type Estimators in the Case of Experiment $1\mathcal{A}$ (Heterogeneous Slopes + Full Rank)

		В	ias (×10	0)		Root Mean Square Errors (×100)						e (5% le	evel, H_0	$: \beta_1 = 1$.00)	Power (5% level, $H_1: \beta_1 = 0.95$)					
CCE T	ype Esti	imators																			
(N,T)	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	
	CCEM	$\mathbf{I}\mathbf{G}$																			
20	0.05	-0.10	-0.03	0.06	-0.07	9.67	7.89	6.74	5.87	5.54	7.20	6.90	7.15	7.90	7.55	11.65	13.00	16.10	17.50	20.10	
30	0.09	-0.01	-0.01	-0.13	0.10	7.69	6.09	5.11	4.54	4.22	6.95	5.30	5.90	6.25	6.35	11.40	14.25	18.05	22.05	26.85	
50	-0.19	0.22	-0.11	0.14	-0.04	5.88	4.61	4.01	3.44	3.13	5.70	5.05	6.65	6.20	5.95	15.10	20.40	25.60	34.10	36.65	
100	0.00	0.04	0.04	0.03	0.04	4.25	3.46	2.89	2.33	2.27	5.75	5.85	5.25	4.90	6.20	23.35	34.30	44.40	56.00	63.25	
200	-0.05	-0.02	-0.03	0.05	0.00	3.07	2.49	2.01	1.72	1.51	4.40	5.15	4.90	5.60	5.10	35.55	52.65	68.70	83.65	90.50	
	CCEP																				
20	0.18	0.00	-0.05	-0.01	-0.13	8.75	7.67	6.85	6.32	6.21	7.70	8.10	7.30	8.05	7.15	12.75	13.50	16.05	16.80	18.30	
30	-0.17	-0.12	0.09	-0.15	0.13	7.10	5.99	5.32	4.78	4.46	7.55	6.25	6.75	6.65	6.45	12.40	15.00	19.30	20.65	26.90	
50	0.00	0.18	-0.07	0.12	-0.01	5.33	4.51	3.97	3.47	3.22	6.80	6.20	5.90	6.35	6.45	17.45	22.15	26.40	32.90	36.25	
100	0.00	0.09	0.03	0.00	0.02	3.78	3.25	2.85	2.34	2.28	5.70	5.65	5.60	5.15	6.25	28.15	37.40	44.80	55.20	61.75	
200	-0.07	-0.04	-0.05	0.05	0.00	2.71	2.29	1.95	1.70	1.53	5.10	4.35	5.05	4.70	4.75	44.75	56.80	70.30	83.55	89.75	
Princip		ponent E	Estimato	rs, Augr	\mathbf{nented}																
	PC1M																				
20	-12.27	-11.15	-10.30	-8.87	-8.90	17.09	14.81	13.24	11.51	11.55	22.55	25.35	30.05	33.40	37.40	12.15	12.95	13.30	12.70	13.75	
30	-9.25	-7.86	-6.46	-5.72	-5.25	13.55	10.84	8.98	7.80	7.15	20.60	20.90	21.65	24.75	24.70	10.75	8.25	7.35	7.40	6.75	
50	-6.84	-5.05	-3.89	-3.01	-3.12	10.10	7.79	5.86	4.67	4.47	19.95	17.65	16.25	14.95	17.90	8.70	8.20	7.65	11.40	9.75	
100	-4.78	-3.21	-2.03	-1.57	-1.45	7.44	5.34	3.68	2.87	2.72	20.10	16.80	11.45	9.75	11.10	9.55	12.15	20.25	28.85	36.75	
200	-4.31	-2.54	-1.39	-0.81	-0.78	6.39	4.19	2.60	1.93	1.71	25.20	17.95	10.95	8.15	7.65	13.85	21.95	42.85	67.65	77.15	
	PC1P																				
20	-11.97	-11.04	-10.35	-9.09	-9.23	15.88	14.38	13.07	11.59	12.07	25.50	28.35	32.05	34.45	38.95	12.05	14.10	14.90	14.55	14.90	
30	-8.86	-7.66	-6.34	-5.73	-5.37	12.48	10.45	8.89	7.80	7.34	21.45	23.75	22.05	24.70	25.50	11.00	8.80	7.55	7.95	6.35	
50	-6.20	-4.86	-3.81	-3.07	-3.19	9.06	7.52	5.72	4.73	4.54	21.40	18.75	16.00	16.05	18.90	8.55	9.55	8.10	10.90	9.65	
100	-4.36	-3.00	-2.01	-1.60	-1.49	6.61	5.01	3.61	2.88	2.74	21.05	16.85	11.25	9.35	10.80	11.25	14.55	20.85	27.90	36.30	
200	-3.62	-2.32	-1.36	-0.81	-0.79	5.39	3.81	2.51	1.91	1.73	25.15	17.60	10.50	7.80	7.80	16.35	26.75	45.45	68.00	76.15	
Princip		ponent E	Estimato	rs, Orth	ogonalis	ed										1					
	PC2M		24.04	22.6=	00.44	00.00	20.01	25.00	20.44	20.00	00 50	00.45	04.05	0 = 00	0= 10		=0.05		00.05	00.00	
20	-31.26	-27.06	-24.01	-22.67	-23.11	32.83	28.34	25.00	23.44	23.83	86.50	88.45	91.25	95.20	97.40	74.10	73.95	75.80	82.05	88.20	
30	-25.50	-21.21	-18.27	-16.69	-16.33	26.82	22.25	19.13	17.35	16.92	86.85	87.10	89.10	93.35	95.95	70.15	67.80	66.10	69.25	74.70	
50	-20.65	-16.23	-13.32	-11.41	-10.89	21.68	17.06	13.98	11.95	11.37	90.15	88.35	88.80	89.05	91.70	70.80	60.25	52.20	45.80	46.10	
100	-16.17	-12.44	-9.69	-7.61	-6.60	16.87	12.97	10.18	7.99	7.02	93.65	93.30	89.75	87.50	83.30	72.35	56.20	37.60	19.30	13.60	
200	-14.61 PC2P(-10.78	-8.12	-5.79	-4.59	15.11	11.19	8.45	6.08	4.85	98.95	97.85	95.45	90.75	83.75	79.65	60.20	33.30	10.00	6.75	
20	PC2PC -31.97	-27.47	-24.27	-23.18	-24.19	33.39	28.69	25.23	23.99	24.99	91.00	90.70	93.20	95.55	98.50	80.65	78.60	78.80	83.35	90.45	
30	-31.97	-21.41 -21.51	-24.27 -18.24	-23.18 -16.83	-24.19 -16.75	27.53	$\frac{28.09}{22.48}$	$\frac{25.23}{19.13}$	$\frac{23.99}{17.51}$	$\frac{24.99}{17.37}$	91.00 91.35	90.70	93.20 89.70	93.35	98.50 96.15	78.50	78.60	66.65	83.33 70.65	76.90	
50	-20.32	-21.31 -16.35	-13.17	-10.85 -11.35	-10.75	$\frac{27.33}{22.10}$	17.15	19.13 13.82	11.91	11.48	91.33	90.40	88.95	93.33 88.20	90.13	79.65	63.80	52.95	46.20	48.25	
100	-21.22 -16.77	-10.35 -12.52	-13.17 -9.62	-11.35 -7.55	-10.99 -6.60	17.43	17.15 13.06	13.82 10.11	7.95	7.03	95.05 97.95	90.90 95.05	90.50	86.45	82.30	80.90	60.80	38.10	18.30	$\frac{48.25}{14.25}$	
200	-15.16	-12.52 -10.91	-9.62 -8.00	-7.55 -5.66	-0.00 -4.53	15.67	13.06 11.33	8.34	$\frac{7.95}{5.96}$	4.79	97.95	95.05 98.45	90.50 95.95	89.35	82.50 82.50	80.90 88.65	65.85	38.10 33.35	8.40	6.30	
	-10.10	-10.91	-0.00	-0.00	-4.03	10.07	11.55	0.04	5.90	4.19	99.70	90.40	90.90	o9.55	82.50	00.00	00.00	აა.აა	0.40	0.50	

(Table 1 Continued)

30 sible MG -0.19 -0.14 0.07 0.07 0.04 sible Poo	(includi 50 3 -0.08 0.01 -0.06 0.02 -0.07	$\begin{array}{c} \text{lng } f_{1t} \\ \text{100} \\ \\ 0.15 \\ -0.02 \\ 0.14 \\ 0.00 \\ 0.06 \end{array}$	and f_{2t}) 200 -0.08 0.12 -0.04 0.04	7.21 5.91 4.48	30 6.33 4.95 3.75	50 5.62 4.43	100	200 4.76	20 6.40	30	50	100	200	20	30	50	100	200
30 sible MC -0.19 -0.14 0.07 0.07 0.04	-0.08 0.01 -0.06 0.02 -0.07	0.15 -0.02 0.14 0.00	-0.08 0.12 -0.04	7.21 5.91 4.48	6.33 4.95	5.62	4.98					100	200	20	30	50	100	200
-0.19 -0.14 -0.07 0.07 0.04	-0.08 0.01 -0.06 0.02 -0.07	0.15 -0.02 0.14 0.00	-0.08 0.12 -0.04	7.21 5.91 4.48	6.33 4.95	5.62	4.98					100	200	20	30	50	100	200
-0.19 -0.14 0.07 0.07 0.04	-0.08 0.01 -0.06 0.02 -0.07	-0.02 0.14 0.00	0.12 -0.04	$5.91 \\ 4.48$	4.95			4.76	6.40									ì
-0.14 0.07 0.07 0.04	0.01 -0.06 0.02 -0.07	-0.02 0.14 0.00	0.12 -0.04	$5.91 \\ 4.48$	4.95			4.76	6.40									
0.07 0.07 0.04	-0.06 0.02 -0.07	$0.14 \\ 0.00$	-0.04	4.48		4.43				6.20	6.80	5.95	6.50	12.75	15.35	16.85	19.70	20.40
$0.07 \\ 0.04$	0.02 -0.07	0.00		_	9.75		3.97	3.87	6.50	5.80	6.05	5.30	5.90	16.15	18.05	23.35	25.20	28.80
0.04	-0.07		0.04			3.39	3.09	2.94	6.45	5.25	5.90	5.25	5.20	21.70	27.35	31.45	38.45	40.25
		0.06	0.04	3.16	2.78	2.49	2.15	2.14	5.50	5.15	5.45	4.70	5.45	36.85	46.15	55.10	62.50	66.65
sible Poc		0.00	0.01	2.22	1.93	1.69	1.57	1.44	4.85	5.00	5.00	5.60	4.70	59.15	72.85	82.25	90.40	92.75
0.10		0.00	0.01	7.00	0.00	0.00	7 11	7.40	0.40	0.00	0.00	7.00	F 10	10.70	10.75	1155	1.4.10	10.05
																		12.65
																		16.60
																		21.00
																		32.30
				2.33	2.22	2.20	2.45	2.49	4.95	4.70	4.50	0.80	4.70	50.15	02.10	59.50	59.05	52.20
	nuaing	J_{1t} and	J2t)															
	26.62	20.06	20.60	91 7C	22.07	97 97	41.40	47.04	20.05	22.05	94.05	25 45	21 50	41.00	49 GE	49 50	41.05	38.05
				l														47.15
																		65.05
																		83.20
																		92.05
	21.40	90.03	00.20	21.01	20.11	02.00	00.11	00.01	01.00	00.00	01.00	00.00	01.50	00.10	31.30	32.00	32.30	32.00
	31.27	33.59	34.84	35.30	37.01	42.66	45.42	47.67	42.15	43.65	47.75	45.20	44.50	52.50	52.65	55.95	53.40	51.95
																		64.55
		34.01		35.61		39.05			64.75	67.15		70.35	69.35	73.55			78.65	77.45
28.47	31.30	33.15	34.91	34.39	36.76	39.90	41.79	44.27	75.85	78.90	81.35	79.30	80.15	85.10	86.55	88.05	86.65	86.40
28.32	31.89	33.65	34.11	34.20	36.21	39.63	42.39	42.68	83.45	86.25	87.70	87.40	87.20	89.95	91.90	93.55	92.20	92.20
	-0.13 -0.15 0.07 0.06 cors (exc MG 23.13 25.06 23.91 23.92 24.09 Pooled 26.60 29.39 28.75 28.47	-0.15 0.22 0.07 -0.08 0.07 0.09 0.06 -0.12 cors (excluding) 23.13 26.82 25.06 28.36 23.91 25.65 23.92 26.76 24.09 27.49 26.60 31.27 29.39 32.45 28.75 30.39 28.47 31.30	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	-0.13	-0.13									

Notes: The DGP is $y_{it} = \alpha_{i1}d_{1t} + \beta_{i2}x_{2it} + \gamma_{i1}f_{1t} + \gamma_{i2}f_{2t} + \varepsilon_{it}$ with $\varepsilon_{it} = \rho_{i\varepsilon}\varepsilon_{i,t-1} + \sigma_{i}(1-\rho_{i\varepsilon}^{2})^{1/2}\omega_{it}$, i=1,2,...,[N/2], and $\varepsilon_{it} = \sigma_{i}(1+\theta_{i\varepsilon}^{2})^{-1/2}(\omega_{it}+\theta_{i\varepsilon}\omega_{i,t-1})$, $i=[N/2]+1,...,N, \ \omega_{it} \sim IIDN\ (0,1), \ \sigma_{i}^{2} \sim IIDU\ [0.5,1.5], \ \rho_{i\varepsilon} \sim IIDU\ [0.05,0.95], \ \theta_{i\varepsilon} \sim IIDU\ [0.01].$ Regressors are generated by $x_{ijt} = a_{ij1}d_{1t} + a_{ij2}d_{2t} + \gamma_{ij1}f_{1t} + \gamma_{ij3}f_{3t} + v_{ijt}$, j=1,2, for i=1,2,...,N. $d_{1t}=1, d_{2t}=0.5d_{2,t-1}+v_{dt}, v_{dt} \sim IIDN\ (0,1-0.5^{2}), \ d_{2,-50}=0$; $f_{jt}=f_{jt-1}+v_{fj,t}, v_{fj,t} \sim IIDN\ (0,1), \ f_{j,-50}=0$, for j=1,2,3; $v_{ijt}=\rho_{vij}v_{ijt-1}+v_{ijt}$, $v_{ijt} \sim IIDN\ (0,1-\rho_{vij}^{2}), v_{ij,-50}=0$ and $\rho_{vij} \sim IIDU\ [0.05,0.95]$ for j=1,2, for t=-49,...,T with the first 50 observations discarded; $\alpha_{i1} \sim IIDN\ (1,1)$; $a_{ij\ell} \sim IIDN\ (0,5,0.50), \ \gamma_{i13}$ and $\gamma_{i23} \sim IIDN\ (0.5,0.50), \ \gamma_{i13}$ and $\gamma_{i23} \sim IIDN\ (0.5,0.50), \ \gamma_{i13}$ and $\gamma_{i23} \sim IIDN\ (0.5,0.50), \ \gamma_{i13}$ and $\gamma_{i23} \sim IIDN\ (0.0.50), \ \gamma_{i13} \sim IIDN\ (0.0.50),$

Table 2: Small Sample Properties of Common Correlated Effects Type Estimators in the Case of Experiment 2A (Homogeneous Slopes + Full Rank)

CCE Type Estimators			В	ias (×10	0)		Root Mean Square Errors (×100)						e (5% l	evel, H_0	$\beta_1 = 1$	1.00)	Power (5% level, $H_1: \beta_1 = 0.95$)					
CCEMB CO	CCE T	ype Esti	$_{ m imators}$,									•			<u> </u>		`			· · ·	
Color Colo	(N,T)	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	
30		CCEM	IG																			
50	1 1																				47.90	
100 -0.04 -0.08 0.06 -0.04 -0.01 3.59 2.76 2.02 1.35 0.98 4.55 5.50 6.05 5.10 6.10 28.30 43.00 72.35 95.20 9 200 0.06 -0.02 0.03 0.01 0.00 2.83 2.05 1.52 1.00 0.08 5.60 4.45 6.35 5.20 5.70 44.20 67.95 91.90 99.90 100 200 0.01 0.02 0.03 0.01 0.02 0.05 5.56 4.94 3.98 3.74 6.60 6.75 7.30 6.75 6.80 14.25 16.25 25.25 33.70 4 30 -0.14 0.14 0.07 0.01 0.01 5.20 4.50 3.55 2.67 2.26 5.10 5.90 7.25 6.25 6.40 15.25 24.55 34.90 52.95 7 50 0.05 0.07 -0.02 0.04 0.03 4.08 3.29 2.66 1.84 1.39 5.40 5.40 5.45 6.20 5.30 24.60 34.35 51.70 78.65 9 200 0.07 -0.03 0.01 0.02 0.00 2.17 1.63 1.32 0.92 0.65 5.60 3.95 5.70 5.60 5.35 65.25 84.40 96.95 100.00 10 Principal Component Estimators, Augmented 20 1.24 -1.13 -1.06 -9.44 -8.84 12.66 12.48 12.18 11.24 10.92 24.55 32.55 3.50 58.50 71.95 13.05 15.30 13.80 19.00 2 30 -9.35 -7.83 -6.39 -5.66 -5.34 12.96 10.55 81.8 6.93 6.15 22.10 22.60 22.00 16.50 6.85 5.85 5.85 1.30 19.00 2.10							-														71.60	
Correct Corr	1 1																				95.00	
CCEP 20 0.18 0.00 0.03 -0.14 0.08 6.95 5.56 4.94 3.98 3.74 6.60 6.75 7.30 6.75 6.80 14.25 16.25 25.25 33.70 4 30 -0.14 0.14 0.07 0.01 0.01 5.20 4.50 3.55 2.67 2.26 5.10 5.90 7.25 6.25 6.40 15.25 24.55 34.90 52.95 7 50 0.05 0.07 -0.02 0.04 0.06 -0.04 -0.01 2.87 2.37 1.78 1.24 0.93 5.00 5.40 5.45 6.20 5.30 24.60 34.35 51.70 78.65 9 100 -0.02 -0.04 0.06 -0.04 -0.01 2.87 2.37 1.78 1.24 0.93 5.00 5.00 6.20 6.40 5.25 5.95 41.65 58.35 81.85 97.80 10 200 0.07 -0.03 0.01 0.02 0.00 2.17 1.63 1.32 0.92 0.65 5.60 3.95 5.70 5.60 5.35 65.25 84.40 96.95 100.00 10 Principal Component Estimators, Nameure PCING 20 -12.34 -11.39 -10.06 -9.44 -8.84 16.76 14.48 12.18 11.24 10.92 24.55 32.55 39.50 58.50 71.95 13.05 15.30 13.80 19.00 2 30 -9.35 -7.83 -6.39 -5.66 -5.34 12.96 10.55 8.18 6.93 6.15 22.10 26.10 32.55 46.60 68.25 9.55 10.40 9.10 9.45 20.80 10.25 1 50 -7.05 -5.28 -3.81 -3.08 -3.17 10.12 7.38 5.11 3.86 3.73 23.55 24.05 25.30 35.65 56.85 10.40 9.10 9.45 20.80 13.10 -5.00 -3.45 -2.04 -1.04 -1.57 7.19 5.16 3.14 2.20 1.90 1.00 16.50 22.45 35.70 9.90 14.45 31.05 43.10 43.10 43.10 10 -5.00 -3.45 -2.04 -1.04 -1.57 7.19 5.16 31.4 2.20 1.90 1.00 16.50 22.45 35.70 9.90 14.45 31.05 40.90 10 10 10 10 10 10 10 10 10 10 10 10 10	1 1																				99.90	
20	200		-0.02	0.03	0.01	0.00	2.83	2.05	1.52	1.00	0.68	5.60	4.45	6.35	5.20	5.70	44.20	67.95	91.90	99.90	100.00	
30																						
The color The	1 1	L																			46.25	
100		l.																			70.70	
Principal Component Estimators, Augmented PCIMC Principal Component Estimators, Augmented PCIMC PCIM	1 1																				95.00	
Principal Component Estimators, Augmented PCIMG 20 -12.34 -11.39 -10.06 -9.44 -8.84 16.76 14.48 12.18 11.24 10.92 24.55 32.55 39.50 58.50 71.95 13.05 15.30 13.80 19.00 2 30 -9.35 -7.83 -6.39 -5.66 -5.34 12.96 10.55 8.18 6.93 6.15 22.10 26.10 32.55 46.60 68.25 9.55 10.60 8.40 10.25 1 50 -7.05 -5.28 -3.81 -3.08 -3.17 10.12 7.38 5.11 3.86 3.73 23.55 24.05 25.30 35.65 56.85 10.40 9.10 9.45 20.80 3 100 -5.00 -3.45 -2.04 -1.64 -1.57 7.19 5.16 3.14 2.20 1.90 22.60 22.00 16.50 22.45 35.70 9.90 14.45 31.05 64.90 9 200 -4.23 -2.65 -1.27 -0.87 -0.79 6.27 4.11 2.13 1.37 1.06 28.05 22.90 14.90 16.85 21.95 16.15 30.55 67.35 97.55 10 PCIPOL 20 -11.78 -11.12 -9.89 -9.43 -8.93 15.09 13.70 11.86 11.20 10.77 28.20 37.15 46.35 64.10 76.75 14.80 17.20 17.65 21.80 2 30 -8.55 -7.35 -6.10 -5.58 -5.35 11.37 9.99 7.66 6.79 6.18 25.60 29.35 35.60 49.60 71.00 10.65 9.95 8.95 10.10 1 3 3 3 3 3 3 3 3 3	1 1	I.										I.					I.				100.00	
PC1MG	1						2.17	1.63	1.32	0.92	0.65	5.60	3.95	5.70	5.60	5.35	65.25	84.40	96.95	100.00	100.00	
20	Princip			Estimato	rs, Aug	$_{ m mented}$																
30																						
The color The	1 1																				21.20	
100		I.																			10.10	
Color Colo	1 1	I.																			30.00	
PC1POOL 20	1 1																l.				91.50	
20	200	l		-1.27	-0.87	-0.79	6.27	4.11	2.13	1.37	1.06	28.05	22.90	14.90	16.85	21.95	16.15	30.55	67.35	97.55	100.00	
30	90			0.00	0.49	0.00	15.00	10.70	11.00	11.00	10.77	00.00	07.15	10.05	04.10	70.75	14.00	17.00	17.05	01.00	04 55	
50 -6.39 -4.86 -3.74 -3.05 -3.19 8.82 6.71 4.87 3.77 3.81 26.60 26.60 28.30 36.55 59.50 10.95 10.25 10.05 22.35 3 100 -4.42 -3.23 -1.98 -1.61 -1.56 6.14 4.68 2.89 2.10 1.87 26.80 25.60 19.70 24.80 38.70 12.65 19.40 40.40 73.05 9 Principal Component Estimators, Orthogonalised PC2MG 20 -31.24 -27.21 -23.95 -22.96 -22.95 32.64 28.30 24.70 23.46 23.37 89.70 92.75 96.35 99.60 100.00 78.30 82.20 85.40 95.90 9 30 -25.74 -21.23 -18.28 -16.52 26.93 22.12 18.86 16.89 16.81 90.55 93.60 97.60 99.60 100.00 78.20 76.50 <th></th> <th>24.55</th>																					24.55	
100	1 1																				10.30 31.90	
200 -3.57 -2.37 -1.21 -0.84 -0.78 5.19 3.57 1.93 1.30 1.03 32.05 25.30 16.60 17.35 24.10 24.55 42.80 79.25 98.80 10 Principal Component Estimators, Orthogonalised PC2MG 20 -31.24 -27.21 -23.95 -22.96 -22.95 32.64 28.30 24.70 23.46 23.37 89.70 92.75 96.35 99.60 100.00 78.30 82.20 85.40 95.90 9 30 -25.74 -21.23 -18.28 -16.52 -16.52 26.93 22.12 18.86 16.89 16.81 90.55 93.60 97.60 99.60 100.00 78.20 76.50 82.60 90.75 9 50 -20.76 -16.51 -13.40 -11.39 -10.92 21.63 17.17 13.81 11.69 11.12 94.65 95.85 98.35 99.85 100.00 78.65 73.60 70.95 73.80 8 100 -16.31 -12.50 -9.58 -7.70 -6.67 16.92 12.94 9.90 7.88 6.81 96.60 97.65 98.25 99.80 99.95 79.25 68.50 52.60 40.60 3 200 -14.51 -10.80 -8.05 -5.85 -4.57 14.98 11.13 8.28 5.98 4.66 99.50 99.65 99.30 99.85 99.95 83.90 69.50 47.75 14.70 1 PC2POOL 20 -31.95 -27.52 -24.18 -23.50 -24.05 33.04 28.47 24.87 24.02 24.56 95.80 96.50 98.25 99.80 100.00 87.95 87.00 89.80 96.95 9 30 -26.27 -21.47 -18.38 -16.67 -16.96 27.25 22.29 18.91 17.06 17.28 96.35 96.25 98.95 99.75 100.00 86.75 83.60 86.75 92.40 9 50 -21.31 -16.46 -13.29 -11.34 -11.05 22.05 17.04 13.68 11.63 11.26 98.80 98.00 99.00 99.90 100.00 89.35 81.05 76.05 74.65 8 100 -16.95 -12.65 -9.52 -7.62 -6.67 17.50 13.05 9.81 7.80 6.81 99.40 99.40 99.40 99.50 99.95 100.00 90.45 78.30 58.35 41.55 3	1 1	I.																			93.65	
Principal Component Estimators, Orthogonalised PC2MG 20		I.																			100.00	
PC2MG 20								5.57	1.95	1.50	1.05	32.03	20.50	10.00	17.55	24.10	24.55	42.80	19.25	90.00	100.00	
20	Frincip		•	Jacimato	is, Ofth	iogonans	eu					I					I					
30 -25.74 -21.23 -18.28 -16.52 -16.52 26.93 22.12 18.86 16.89 16.81 90.55 93.60 97.60 99.60 100.00 78.20 76.50 82.60 90.75 9 50 -20.76 -16.51 -13.40 -11.39 -10.92 21.63 17.17 13.81 11.69 11.12 94.65 95.85 98.35 99.85 100.00 78.65 73.60 70.95 73.80 8 100 -16.31 -12.50 -9.58 -7.70 -6.67 16.92 12.94 9.90 7.88 6.81 96.60 97.65 98.25 99.80 99.95 79.25 68.50 52.60 40.60 3 200 -14.51 -10.80 -8.05 -5.85 -4.57 14.98 11.13 8.28 5.98 4.66 99.50 99.85 99.95 83.90 69.50 47.75 14.70 1 PC2POOL 20 -	20			-93.05	-22.06	-99.05	32.64	28 30	24.70	23.46	23 37	80.70	02 75	06.35	99.60	100.00	78 30	82 20	85.40	95.90	98.25	
50 -20.76 -16.51 -13.40 -11.39 -10.92 21.63 17.17 13.81 11.69 11.12 94.65 95.85 98.35 99.85 100.00 78.65 73.60 70.95 73.80 8 100 -16.31 -12.50 -9.58 -7.70 -6.67 16.92 12.94 9.90 7.88 6.81 96.60 97.65 98.25 99.80 99.95 79.25 68.50 52.60 40.60 3 200 -14.51 -10.80 -8.05 -5.85 -4.57 14.98 11.13 8.28 5.98 4.66 99.50 99.65 99.30 99.85 99.95 83.90 69.50 47.75 14.70 1 PC2POOL 20 -31.95 -27.52 -24.18 -23.50 -24.05 33.04 28.47 24.87 24.02 24.56 95.80 96.50 98.25 99.80 100.00 87.95 87.00 89.80 96.95 9 </th <th></th> <th>95.25 97.65</th>																					95.25 97.65	
100 -16.31 -12.50 -9.58 -7.70 -6.67 16.92 12.94 9.90 7.88 6.81 96.60 97.65 98.25 99.80 99.95 79.25 68.50 52.60 40.60 3 200 -14.51 -10.80 -8.05 -5.85 -4.57 14.98 11.13 8.28 5.98 4.66 99.50 99.65 99.30 99.85 99.95 83.90 69.50 47.75 14.70 1 PC2POOL 30 -27.52 -24.18 -23.50 -24.05 33.04 28.47 24.87 24.02 24.56 95.80 96.50 98.25 99.80 100.00 87.95 87.00 89.80 96.95 9 30 -26.27 -21.47 -18.38 -16.67 -16.96 27.25 22.29 18.91 17.06 17.28 96.35 96.25 98.95 99.75 100.00 86.75 83.60 86.75 92.40 9 50 -21.31 -16.46 -13.29 -11.34 -11.05 22.05 17.04																					86.75	
200 -14.51 -10.80 -8.05 -5.85 -4.57 14.98 11.13 8.28 5.98 4.66 99.50 99.65 99.30 99.85 99.95 83.90 69.50 47.75 14.70 1	1 1	1																			32.25	
PC2POOL 20 -31.95 -27.52 -24.18 -23.50 -24.05 33.04 28.47 24.87 24.02 24.56 95.80 96.50 98.25 99.80 100.00 87.95 87.00 89.80 96.95 9 30 -26.27 -21.47 -18.38 -16.67 -16.96 27.25 22.29 18.91 17.06 17.28 96.35 96.25 98.95 99.75 100.00 86.75 83.60 86.75 92.40 9 50 -21.31 -16.46 -13.29 -11.34 -11.05 22.05 17.04 13.68 11.63 11.26 98.80 98.00 99.00 99.90 100.00 89.35 81.05 76.05 74.65 8 100 -16.95 -12.65 -9.52 -7.62 -6.67 17.50 13.05 9.81 7.80 6.81 99.40 99.40 99.50 99.95 100.00 90.45 78.30 58.35 41.55 3																					13.00	
20 -31.95 -27.52 -24.18 -23.50 -24.05 33.04 28.47 24.87 24.02 24.56 95.80 96.50 98.25 99.80 100.00 87.95 87.00 89.80 96.95 9 30 -26.27 -21.47 -18.38 -16.67 -16.96 27.25 22.29 18.91 17.06 17.28 96.35 96.25 98.95 99.75 100.00 86.75 83.60 86.75 92.40 9 50 -21.31 -16.46 -13.29 -11.34 -11.05 22.05 17.04 13.68 11.63 11.26 98.80 98.00 99.00 99.90 100.00 89.35 81.05 76.05 74.65 8 100 -16.95 -12.65 -9.52 -7.62 -6.67 17.50 13.05 9.81 7.80 6.81 99.40 99.40 99.50 99.95 100.00 90.45 78.30 58.35 41.55 3				0.00	0.00	1.01	11.00	11.19	0.20	0.00	1.00	30.00	00.00	55.50	30.00	00.00	33.00	00.00	11.15	11.10	10.00	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	20			-24.18	-23.50	-24.05	33.04	28.47	24.87	24.02	24.56	95.80	96.50	98.25	99.80	100.00	87.95	87.00	89.80	96.95	99.30	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		l																			98.35	
100 -16.95 -12.65 -9.52 -7.62 -6.67 17.50 13.05 9.81 7.80 6.81 99.40 99.40 99.50 99.95 100.00 90.45 78.30 58.35 41.55 3																					88.35	
																					32.65	
1 200 1 -15.07 -10.92 -7.95 -5.72 -4.52 1 15.52 11.24 8.15 5.85 4.00 1 99.90 99.90 99.90 99.90 99.90 99.40 80.20 51.05 14.00 1	200	-15.07	-10.92	-7.93	-5.72	-4.52	15.52	11.24	8.15	5.85	4.60	99.90	99.95	99.90	99.90	99.95	94.80	80.20	51.05	14.00	15.30	

Notes: The DGP is the same as that of Table 1, except $\beta_{ij}=1$ for all i and $j,\,i=1,2,...,N,\,j=1,2$. See notes to Table 1.

Table 3: Small Sample Properties of Common Correlated Effects Type Estimators in the Case of Experiment $1\mathcal{B}$ (Heterogeneous Slopes + Rank Deficient)

			as (×100))		Root Mean Square Errors ($\times 100$)						e (5% le	evel, H_0	$: \beta_1 = 1$.00)	Power (5% level, $H_1: \beta_1 = 0.95$)					
	ype Estin																				
(N,T)	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	
	CCEMO																				
20	0.33	-0.19	0.20	0.14	0.23	15.02	13.90	12.61	13.35	13.78	6.80	6.90	6.75	6.60	7.20	9.40	8.95	10.15	10.15	10.15	
30	0.30	0.14	0.09	-0.17	0.35	12.91	12.03	10.70	10.07	10.59	5.50	6.80	5.25	6.15	4.80	8.40	10.05	9.45	10.35	11.65	
50	-0.15	0.63	-0.20	-0.17	0.02	9.82	8.46	7.87	7.42	7.34	5.80	5.10	6.10	5.75	5.90	9.75	12.90	13.40	14.00	15.20	
100	0.25	0.13	0.27	0.00	0.06	7.01	6.55	5.85	5.25	5.01	5.75	5.95	5.45	5.45	6.10	14.50	17.75	21.65	22.65	27.30	
200	0.05	-0.11	-0.17	-0.07	-0.05	5.35	4.65	4.15	3.61	3.31	4.80	5.05	4.75	5.15	4.55	19.45	23.70	29.75	37.25	43.45	
	CCEP								40												
20	0.48	0.06	-0.04	0.16	0.10	13.13	12.81	12.21	13.57	15.30	6.75	7.40	7.00	6.65	6.75	9.90	10.20	10.40	10.35	10.25	
30	-0.23	-0.06	0.18	-0.25	0.43	11.48	10.70	10.39	9.95	11.04	6.10	6.90	5.70	6.00	5.50	9.05	9.95	10.55	10.25	10.60	
50	0.00	0.48	-0.18	-0.17	-0.02	8.42	7.57	7.23	7.22	7.22	5.25	5.90	6.25	5.30	5.50	11.40	14.05	14.15	14.35	15.20	
100	0.11	0.18	0.24	-0.06	0.05	5.87	5.72	5.27	4.87	4.98	5.10	6.00	5.40	4.95	6.00	17.25	19.60	23.50	23.55	27.00	
200	0.04	-0.10	-0.16	-0.04	-0.03	4.35	3.99	3.75	3.30	3.15	5.40	4.70	5.25	4.10	3.95	25.75	28.50	34.50	41.10	46.05	
Princip	al Compo		timator	s, Augm	ented																
	PC1MG														• • • • •						
20	-8.33	-8.02	-7.93	-7.27	-7.44	14.50	12.28	10.98	9.59	9.25	13.35	16.10	21.45	24.50	28.80	7.20	7.35	7.20	7.90	8.00	
30	-5.36	-4.97	-4.70	-4.58	-4.27	10.67	8.66	7.30	6.53	5.96	10.00	11.45	13.80	17.60	18.75	5.85	5.40	5.35	5.40	5.60	
50	-3.15	-2.86	-2.99	-2.59	-2.63	7.44	5.94	5.15	4.33	4.07	7.95	8.25	12.35	11.65	14.30	6.40	6.50	8.70	12.45	11.50	
100	-1.35	-1.27	-1.33	-1.26	-1.17	5.04	4.01	3.24	2.68	2.55	6.80	8.00	7.85	7.40	8.40	13.50	18.90	24.85	31.90	42.30	
200	-0.85	-0.77	-0.72	-0.58	-0.62	3.56	2.83	2.18	1.83	1.64	4.95	6.35	6.05	6.75	6.30	23.40	36.25	53.50	72.15	80.20	
	PC1PO		7.07	7.00	7.00	10.04	11 50	10.50	0.40	0.14	10.45	10.70	00.05	05.00	00.05	7 50	7.00	7.00	F 05	7.05	
20	-7.93	-7.85	-7.87	-7.23	-7.32	12.94	11.59	10.53	9.43	9.14	13.45	16.70	22.35	25.30	28.35	7.50	7.30	7.60	7.65	7.35	
30	-5.33	-4.94	-4.61 -2.92	-4.54	-4.21	9.81	8.23	7.13	6.50	5.92	11.10	12.55	13.25	17.70	18.40	6.55	5.90	5.60	5.70	5.35	
50	-2.98	-2.78		-2.63	-2.65	6.62	5.67	5.01	4.37	4.12	8.00	9.85	11.35	12.15	14.20	7.40	7.90	8.40	11.70	11.50	
100 200	-1.41 -0.82	-1.23	-1.36	-1.31	-1.21 -0.64	$4.37 \\ 3.02$	3.66	3.19	2.71	2.57	6.80	7.25	7.50	7.70	9.15	15.85	21.20	26.30	31.30	40.80	
	al Compo	-0.77	-0.75	-0.60			2.57	2.12	1.82	1.67	6.05	6.35	6.65	5.90	6.75	29.20	42.25	56.25	71.40	78.95	
Princip	PC2MG		timator	s, Ortno	gonanse	a															
20	-30.74	-26.62	-23.99	-22.56	-23.18	32.50	27.94	25.06	23.36	23.92	82.50	87.55	90.45	95.10	97.55	70.50	72.50	74.75	81.15	87.80	
30	-30.74 -24.89	-20.02 -20.85	-23.99 -18.19	-22.50 -16.68	-23.18 -16.43	$\frac{32.30}{26.28}$	$\frac{27.94}{21.96}$	19.03	$\frac{25.30}{17.35}$	$\frac{25.92}{17.03}$	84.15	86.05	90.45 90.35	93.10 93.60	97.55 96.05	67.80	65.45	66.75	69.85	75.65	
50	-24.69	-20.65 -15.65	-13.13	-10.08	-10.45	20.28	16.50	13.80	17.33 11.93	17.03 11.34	87.30	86.20	90.33 87.95	89.65	90.03 91.70	65.10	57.00	51.20	45.00	46.35	
100	-15.19	-11.94	-13.13 -9.58	-11.56 -7.57	-6.60	15.96	12.53	10.09	7.96	7.03	91.80	90.95	88.90	87.35	82.65	65.05	50.60	37.45	18.45	14.20	
200	-13.13	-11.34	-7.98	-7.37 -5.75	-4.58	14.16	12.93 10.80	8.33	6.05	4.85	98.50	96.95	95.00	90.60	83.65	72.00	55.40	31.15	9.30	6.80	
200	PC2PO		-1.30	-0.10	-4.00	14.10	10.00	0.00	0.00	4.00	30.00	<i>3</i> 0. <i>3</i> 0	30.00	30.00	00.00	12.00	99.40	91.19	g. 9 0	0.00	
20	-31.30	-27.08	-24.30	-23.16	-24.26	32.81	28.29	25.32	23.99	25.10	88.95	89.95	93.15	96.00	98.75	79.05	77.70	79.10	83.00	90.25	
30	-25.55	-21.11	-18.14	-16.84	-16.86	26.80	$\frac{26.29}{22.15}$	18.99	17.53	17.49	90.25	88.25	90.40	94.00	96.30	75.35	68.95	66.20	70.00	77.15	
50	-19.99	-21.11	-13.14	-10.34	-10.96	20.92	16.56	13.65	11.91	11.46	93.30	89.05	88.30	88.65	91.70	74.15	60.30	51.40	45.55	47.50	
100	-15.72	-11.98	-13.00 -9.51	-7.52	-6.60	16.43	12.56	10.01	7.91	7.03	96.00	92.55	89.40	86.50	81.95	74.13	54.45	36.55	17.60	14.00	
200	-14.09	-11.33	-7.86	-5.63	-4.52	14.61	10.90	8.21	5.93	4.79	99.30	97.95	95.20	88.75	81.90	82.55	60.20	32.10	8.25	6.15	
200	-14.09	-10.47	-1.00	-0.00	-4.02	14.01	10.90	0.21	ნ.ჟმ	4.13	<i>33.</i> 30	31.30	30.20	00.10	31.90	04.00	00.20	32.10	0.20	0.10	

Notes: The DGP is the same as that of Table 1, except $\gamma_{i2} \sim IIDN$ (0,1), so that the rank condition is not satisfied. See notes to Table 1.

Table 4: Small Sample Properties of Common Correlated Effects Type Estimators in the Case of Experiment $2\mathcal{B}$ (Homogeneous Slopes + Rank Deficient)

		В	ias (×100	0)		Root Mean Square Errors (×100)						e (5% l	evel, H_0	$\beta_1 = 1$	1.00)	Power (5% level, $H_1: \beta_1 = 0.95$)					
CCE T	ype Esti	mators	`	,					`	,		`	, ,		,			,	. 1	,	
(N,T)	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	20	30	50	100	200	
	CCEM	\mathbf{G}																			
20	-0.28	-0.26	0.41	-0.31	0.73	14.45	12.85	12.02	12.07	13.47	7.35	5.45	6.40	6.70	6.00	9.35	9.15	10.95	11.55	10.90	
30	-0.11	0.07	0.09	0.45	-0.05	11.99	10.78	9.82	9.52	10.33	5.20	5.90	5.95	6.50	6.55	7.85	10.50	12.40	14.35	14.90	
50	0.00	0.23	-0.07	-0.02	0.00	9.01	7.97	7.62	6.79	6.72	5.05	4.80	5.00	5.45	4.95	9.40	12.20	15.75	17.60	21.15	
100	0.14	-0.08	-0.12	-0.03	0.06	6.66	5.92	5.16	4.78	4.56	4.65	5.40	5.60	4.60	6.35	15.10	18.15	23.95	28.50	34.85	
200	0.14	0.11	0.01	-0.17	-0.07	5.13	4.45	3.88	3.27	3.34	5.45	5.10	5.45	4.65	5.15	22.35	28.80	36.60	44.75	56.70	
	CCEP																				
20	-0.12	-0.19	0.35	-0.26	0.66	12.66	11.53	11.56	12.12	15.07	7.45	7.00	7.55	6.35	6.50	9.85	10.00	12.60	12.65	11.50	
30	-0.09	0.05	0.06	0.39	0.03	10.00	9.57	9.26	9.36	11.05	5.55	5.75	6.80	6.70	6.75	9.90	11.70	13.30	15.20	14.50	
50	-0.14	0.39	-0.08	0.01	0.03	7.29	6.92	6.84	6.58	6.79	4.95	5.25	5.45	5.60	4.85	11.25	15.60	16.65	19.95	20.40	
100	0.20	-0.13	-0.11	-0.05	0.04	5.44	4.97	4.55	4.45	4.39	4.80	5.35	5.40	4.95	6.05	20.60	22.65	28.35	31.40	36.80	
200	0.19	0.11	-0.08	-0.13	-0.07	3.97	3.71	3.35	2.96	3.09	5.25	5.15	5.05	5.00	5.60	31.95	38.45	44.30	50.70	60.40	
Princip	al Comp		stimato	rs, Augı	\mathbf{mented}																
	PC1M																				
20	-8.22	-8.36	-7.93	-7.89	-7.48	13.75	11.78	10.07	9.07	8.40	13.30	18.95	27.70	46.00	58.60	7.25	7.80	8.20	9.55	9.40	
30	-5.40	-5.01	-4.70	-4.59	-4.52	10.11	8.07	6.49	5.54	5.03	10.35	13.45	19.70	33.45	52.05	5.95	5.45	6.45	5.80	4.95	
50	-3.24	-3.09	-2.81	-2.66	-2.66	6.96	5.35	4.22	3.39	3.04	9.40	11.05	15.90	25.40	44.85	6.85	7.80	11.60	22.45	35.50	
100	-1.53	-1.45	-1.28	-1.33	-1.28	4.54	3.40	2.57	1.96	1.64	6.10	7.75	10.25	15.80	26.30	13.10	20.95	41.60	71.75	95.70	
200	-0.80	-0.76	-0.67	-0.64	-0.63	3.40	2.39	1.74	1.22	0.94	6.00	5.45	7.75	10.75	15.70	27.15	46.80	79.60	98.65	100.00	
	PC1PC		7 -7	7.01	7.10	11 45	10.00	0.20	0.50	7.01	14.05	01.70	21.00	40.05	F7 0F	C 40	0.95	7	0.00	0.05	
20 30	-7.56 -5.00	-7.83 -4.76	-7.57 -4.50	-7.61 -4.43	-7.12 -4.35	$11.45 \\ 8.47$	$10.60 \\ 7.20$	$9.32 \\ 6.04$	$8.59 \\ 5.25$	$7.81 \\ 4.82$	14.05 11.70	$21.70 \\ 15.75$	$31.20 \\ 21.45$	$48.25 \\ 35.10$	57.85 51.40	6.40 6.00	8.35 5.70	$7.55 \\ 6.50$	$8.90 \\ 5.65$	8.35 5.40	
50	-5.00 -2.89	-4.76 -2.77	-4.50 -2.75	-4.43 -2.60	-4.35 -2.61	5.66	$\frac{7.20}{4.63}$	$\frac{6.04}{3.92}$	$\frac{5.25}{3.26}$	$\frac{4.82}{2.97}$	8.90	15.75 11.55	$\frac{21.45}{17.50}$	$\frac{35.10}{27.45}$	44.70	7.85	9.60	12.95	$\frac{5.65}{24.65}$	$\frac{5.40}{38.05}$	
100	-2.89 -1.38	-2.77 -1.39	-2.75 -1.26	-2.00 -1.31	-2.01	$\frac{3.66}{3.67}$	$\frac{4.05}{2.95}$	$\frac{3.92}{2.29}$	$\frac{3.20}{1.87}$	1.60	7.80	8.70	17.50 10.50	17.90	28.50	19.90	$\frac{9.60}{28.70}$	51.45	79.70	97.35	
200	-1.36 -0.72	-1.59 -0.74	-1.20 -0.67	-0.63	-0.63	$\frac{3.07}{2.54}$	$\frac{2.95}{1.93}$	$\frac{2.29}{1.55}$	1.14	0.91	6.25	5.90	9.00	11.80	$\frac{26.50}{17.10}$	43.25	64.40	89.05	99.45	100.00	
					ogonalis		1.90	1.00	1.14	0.91	0.20	9.90	9.00	11.00	17.10	40.20	04.40	09.00	99.40	100.00	
Timeip	PC2M		istillato.	is, Oitil	ogonans	eu															
20	-30.58	-26.66	-23.75	-22.85	-23.07	32.14	27.74	24.51	23.36	23.51	87.40	91.15	95.95	99.60	99.95	75.80	79.65	85.50	94.95	98.85	
30	-25.10	-20.76	-18.15	-16.58	-16.60	26.40	21.79	18.75	16.97	16.90	89.25	92.45	97.20	99.75	100.00	74.35	74.15	80.85	90.90	98.10	
50	-19.66	-15.83	-13.16	-11.36	-10.91	20.58	16.50	13.60	11.67	11.12	92.45	94.35	97.90	99.75	100.00	72.90	69.30	69.35	72.45	86.90	
100	-15.37	-11.97	-9.44	-7.65	-6.66	16.01	12.43	9.75	7.84	6.80	95.30	96.70	98.00	99.70	99.95	74.25	63.40	50.10	40.00	32.15	
200	-13.52	-10.36	-7.93	-5.82	-4.57	14.01	10.70	8.17	5.96	4.66	98.80	99.15	99.40	99.90	99.95	76.85	63.50	45.30	14.80	13.35	
	PC2PC																				
20	-31.34	-27.06	-24.05	-23.47	-24.10	32.52	28.01	24.76	24.00	24.63	95.25	95.90	97.85	99.95	100.00	86.20	86.30	89.00	97.50	99.40	
30	-25.58	-20.90	-18.28	-16.76	-17.05	26.61	21.72	18.83	17.15	17.38	95.30	95.80	98.85	99.85	100.00	84.80	81.60	85.35	93.05	98.50	
50	-20.12	-15.76	-13.06	-11.33	-11.05	20.90	16.33	13.46	11.63	11.26	97.65	97.50	98.90	99.75	100.00	85.10	77.55	73.60	75.00	88.75	
100	-15.88	-12.09	-9.36	-7.58	-6.66	16.46	12.51	9.65	7.76	6.80	99.10	99.00	99.45	99.85	99.95	86.15	72.45	55.15	39.85	32.95	
200	-13.95	-10.45	-7.81	-5.69	-4.52	14.41	10.78	8.03	5.82	4.60	99.70	99.85	99.75	99.85	100.00	90.10	74.75	49.05	13.20	15.30	

Notes: The DGP is the same as that of Table 1, except $\gamma_{i2} \sim IIDN$ (0, 1), so that the rank condition is not satisfied, and $\beta_{ij} = 1$ for all i and j, i = 1, 2, ..., N, j = 1, 2. See notes to Table 1.

CESifo Working Paper Series

(for full list see www.cesifo-group.de)

- 1724 Axel Dreher and Jan-Egbert Sturm, Do IMF and World Bank Influence Voting in the UN General Assembly?, May 2006
- 1725 Swapan K. Bhattacharya and Biswa N. Bhattacharyay, Prospects of Regional Cooperation in Trade, Investment and Finance in Asia: An Empirical Analysis on BIMSTEC Countries and Japan, May 2006
- 1726 Philippe Choné and Laurent Linnemer, Assessing Horizontal Mergers under Uncertain Efficiency Gains, May 2006
- 1727 Daniel Houser and Thomas Stratmann, Selling Favors in the Lab: Experiments on Campaign Finance Reform, May 2006
- 1728 E. Maarten Bosker, Steven Brakman, Harry Garretsen and Marc Schramm, A Century of Shocks: The Evolution of the German City Size Distribution 1925 1999, May 2006
- 1729 Clive Bell and Hans Gersbach, Growth and Enduring Epidemic Diseases, May 2006
- 1730 W. Bentley MacLeod, Reputations, Relationships and the Enforcement of Incomplete Contracts, May 2006
- 1731 Jan K. Brueckner and Ricardo Flores-Fillol, Airline Schedule Competition: Product-Quality Choice in a Duopoly Model, May 2006
- 1732 Kerstin Bernoth and Guntram B. Wolff, Fool the Markets? Creative Accounting, Fiscal Transparency and Sovereign Risk Premia, May 2006
- 1733 Emmanuelle Auriol and Pierre M. Picard, Government Outsourcing: Public Contracting with Private Monopoly, May 2006
- 1734 Guglielmo Maria Caporale and Luis A. Gil-Alana, Modelling Structural Breaks in the US, UK and Japanese Unemployment Rates, May 2006
- 1735 Emily J. Blanchard, Reevaluating the Role of Trade Agreements: Does Investment Globalization Make the WTO Obsolete?, May 2006
- 1736 Per Engström and Bertil Holmlund, Tax Evasion and Self-Employment in a High-Tax Country: Evidence from Sweden, May 2006
- 1737 Erkki Koskela and Mikko Puhakka, Cycles and Indeterminacy in Overlapping Generations Economies with Stone-Geary Preferences, May 2006
- 1738 Saku Aura and Thomas Davidoff, Supply Constraints and Housing Prices, May 2006

- 1739 Balázs Égert and Ronald MacDonald, Monetary Transmission Mechanism in Transition Economies: Surveying the Surveyable, June 2006
- 1740 Ben J. Heijdra and Ward E. Romp, Ageing and Growth in the Small Open Economy, June 2006
- 1741 Robert Fenge and Volker Meier, Subsidies for Wages and Infrastructure: How to Restrain Undesired Immigration, June 2006
- 1742 Robert S. Chirinko and Debdulal Mallick, The Elasticity of Derived Demand, Factor Substitution and Product Demand: Corrections to Hicks' Formula and Marshall's Four Rules, June 2006
- 1743 Harry P. Bowen, Haris Munandar and Jean-Marie Viaene, Evidence and Implications of Zipf's Law for Integrated Economies, June 2006
- 1744 Markku Lanne and Helmut Luetkepohl, Identifying Monetary Policy Shocks via Changes in Volatility, June 2006
- 1745 Timo Trimborn, Karl-Josef Koch and Thomas M. Steger, Multi-Dimensional Transitional Dynamics: A Simple Numberical Procedure, June 2006
- 1746 Vivek H. Dehejia and Yiagadeesen Samy, Labor Standards and Economic Integration in the European Union: An Empirical Analysis, June 2006
- 1747 Carlo Altavilla and Paul De Grauwe, Forecasting and Combining Competing Models of Exchange Rate Determination, June 2006
- 1748 Olaf Posch and Klaus Waelde, Natural Volatility, Welfare and Taxation, June 2006
- 1749 Christian Holzner, Volker Meier and Martin Werding, Workfare, Monitoring, and Efficiency Wages, June 2006
- 1750 Steven Brakman, Harry Garretsen and Charles van Marrewijk, Agglomeration and Aid, June 2006
- 1751 Robert Fenge and Jakob von Weizsäcker, Mixing Bismarck and Child Pension Systems: An Optimum Taxation Approach, June 2006
- 1752 Helge Berger and Michael Neugart, Labor Courts, Nomination Bias, and Unemployment in Germany, June 2006
- 1753 Chris van Klaveren, Bernard van Praag and Henriette Maassen van den Brink, A Collective Household Model of Time Allocation a Comparison of Native Dutch and Immigrant Households in the Netherlands, June 2006
- 1754 Marko Koethenbuerger, Ex-Post Redistribution in a Federation: Implications for Corrective Policy, July 2006

- 1755 Axel Dreher, Jan-Egbert Sturm and Heinrich Ursprung, The Impact of Globalization on the Composition of Government Expenditures: Evidence from Panel Data, July 2006
- 1756 Richard Schmidtke, Private Provision of a Complementary Public Good, July 2006
- 1757 J. Atsu Amegashie, Intentions and Social Interactions, July 2006
- 1758 Alessandro Balestrino, Tax Avoidance, Endogenous Social Norms, and the Comparison Income Effect, July 2006
- 1759 Øystein Thøgersen, Intergenerational Risk Sharing by Means of Pay-as-you-go Programs an Investigation of Alternative Mechanisms, July 2006
- 1760 Pascalis Raimondos-Møller and Alan D. Woodland, Steepest Ascent Tariff Reforms, July 2006
- 1761 Ronald MacDonald and Cezary Wojcik, Catching-up, Inflation Differentials and Credit Booms in a Heterogeneous Monetary Union: Some Implications for EMU and new EU Member States, July 2006
- 1762 Robert Dur, Status-Seeking in Criminal Subcultures and the Double Dividend of Zero-Tolerance, July 2006
- 1763 Christa Hainz, Business Groups in Emerging Markets Financial Control and Sequential Investment, July 2006
- 1764 Didier Laussel and Raymond Riezman, Fixed Transport Costs and International Trade, July 2006
- 1765 Rafael Lalive, How do Extended Benefits Affect Unemployment Duration? A Regression Discontinuity Approach, July 2006
- 1766 Eric Hillebrand, Gunther Schnabl and Yasemin Ulu, Japanese Foreign Exchange Intervention and the Yen/Dollar Exchange Rate: A Simultaneous Equations Approach Using Realized Volatility, July 2006
- 1767 Carsten Hefeker, EMU Enlargement, Policy Uncertainty and Economic Reforms, July 2006
- 1768 Giovanni Facchini and Anna Maria Mayda, Individual Attitudes towards Immigrants: Welfare-State Determinants across Countries, July 2006
- 1769 Maarten Bosker and Harry Garretsen, Geography Rules Too! Economic Development and the Geography of Institutions, July 2006
- 1770 M. Hashem Pesaran and Allan Timmermann, Testing Dependence among Serially Correlated Multi-category Variables, July 2006
- 1771 Juergen von Hagen and Haiping Zhang, Financial Liberalization in a Small Open Economy, August 2006

- 1772 Alessandro Cigno, Is there a Social Security Tax Wedge?, August 2006
- 1773 Peter Egger, Simon Loretz, Michael Pfaffermayr and Hannes Winner, Corporate Taxation and Multinational Activity, August 2006
- 1774 Jeremy S.S. Edwards, Wolfgang Eggert and Alfons J. Weichenrieder, The Measurement of Firm Ownership and its Effect on Managerial Pay, August 2006
- 1775 Scott Alan Carson and Thomas N. Maloney, Living Standards in Black and White: Evidence from the Heights of Ohio Prison Inmates, 1829 1913, August 2006
- 1776 Richard Schmidtke, Two-Sided Markets with Pecuniary and Participation Externalities, August 2006
- 1777 Ben J. Heijdra and Jenny E. Ligthart, The Transitional Dynamics of Fiscal Policy in Small Open Economies, August 2006
- 1778 Jay Pil Choi, How Reasonable is the 'Reasonable' Royalty Rate? Damage Rules and Probabilistic Intellectual Property Rights, August 2006
- 1779 Ludger Woessmann, Efficiency and Equity of European Education and Training Policies, August 2006
- 1780 Gregory Ponthiere, Growth, Longevity and Public Policy, August 2006
- 1781 Laszlo Goerke, Corporate and Personal Income Tax Declarations, August 2006
- 1782 Florian Englmaier, Pablo Guillén, Loreto Llorente, Sander Onderstal and Rupert Sausgruber, The Chopstick Auction: A Study of the Exposure Problem in Multi-Unit Auctions, August 2006
- 1783 Adam S. Posen and Daniel Popov Gould, Has EMU had any Impact on the Degree of Wage Restraint?, August 2006
- 1784 Paolo M. Panteghini, A Simple Explanation for the Unfavorable Tax Treatment of Investment Costs, August 2006
- 1785 Alan J. Auerbach, Why have Corporate Tax Revenues Declined? Another Look, August 2006
- 1786 Hideshi Itoh and Hodaka Morita, Formal Contracts, Relational Contracts, and the Holdup Problem, August 2006
- 1787 Rafael Lalive and Alejandra Cattaneo, Social Interactions and Schooling Decisions, August 2006
- 1788 George Kapetanios, M. Hashem Pesaran and Takashi Yamagata, Panels with Nonstationary Multifactor Error Structures, August 2006